

### **Copyright Undertaking**

This thesis is protected by copyright, with all rights reserved.

#### By reading and using the thesis, the reader understands and agrees to the following terms:

- 1. The reader will abide by the rules and legal ordinances governing copyright regarding the use of the thesis.
- 2. The reader will use the thesis for the purpose of research or private study only and not for distribution or further reproduction or any other purpose.
- 3. The reader agrees to indemnify and hold the University harmless from and against any loss, damage, cost, liability or expenses arising from copyright infringement or unauthorized usage.

#### IMPORTANT

If you have reasons to believe that any materials in this thesis are deemed not suitable to be distributed in this form, or a copyright owner having difficulty with the material being included in our database, please contact <a href="https://www.lbsys@polyu.edu.hk">lbsys@polyu.edu.hk</a> providing details. The Library will look into your claim and consider taking remedial action upon receipt of the written requests.

Pao Yue-kong Library, The Hong Kong Polytechnic University, Hung Hom, Kowloon, Hong Kong

http://www.lib.polyu.edu.hk

# MODELLING INDIVIDUAL AND HOUSEHOLD ACTIVITY-TRAVEL SCHEDULING BEHAVIOURS IN STOCHASTIC TRANSPORTATION NETWORKS

## XIONG YILIANG

M.Phil

The Hong Kong Polytechnic University

2015

## The Hong Kong Polytechnic University

## Department of Civil and Environmental Engineering

# Modelling Individual and Household Activity-Travel Scheduling Behaviours in Stochastic Transportation Networks

Xiong Yiliang

A thesis submitted in partial fulfilment of the requirements for the degree of Master of Philosophy

May 2013

## **Certificate of Originality**

I hereby declare that this thesis is my own work and that, to the best of my knowledge and belief, it reproduces no material previously published or written, nor material that has been accepted for the award of any other degree or diploma, except where due acknowledgement has been made in the text.

\_\_\_\_\_(Signed)

Xiong Yiliang (Name of student)

#### Abstract

The fundamental objective of transportation planning is to provide appropriate transportation facilities to meet future travel demand. Knowledge of travel demand can help improve the efficiency and sustainability of the existing transportation systems. Various approaches, such as network equilibrium models and simulation systems, have been proposed to predict travel demand. The activity-based network equilibrium approach for analysing travel behaviour and predicting travel demand has emerged over the last two decades. The activity-based network equilibrium approach covers a new class of models that predict where and when activities are pursued.

The main contribution of this thesis is the development of a modelling framework for the individual and household daily activity-travel scheduling behaviour. The existing activity-based network equilibrium modelling methodologies are extended in two directions: (1) developing a dynamic model for individual's daily activitytravel scheduling behaviour in congested networks, (2) taking into account the impact of intra-household interactions on daily activity-travel scheduling.

In previous activity-based network equilibrium models, travellers are assumed to care only about the utility that can be obtained immediately. Travellers would thus choose the activity that provides the highest immediate utility but would ignore the utility that could be obtained during the remainder of a day. The daily activity-travel schedule is thus composed of a sequence of repeated static choices over an entire day. In this thesis, a Markov Decision Process (MDP) model is proposed to capture two types of within-day dynamics in individual's activity-travel scheduling behaviour. Firstly, travellers take into account the expected future utility and perceive the impact of the current decision on subsequent activities and trips.

Secondly, activity-travel decisions have a strong dependency on contextual situations, such as time of the day and location.

The impact of travel time uncertainty on activity-travel scheduling is also considered in the proposed MDP model. For example, a sudden increase in travel time to the destination of a non-compulsory activity may reduce the probability of choosing this activity. The MDP model provides a framework for modelling intertemporal activity-travel decisions in such an uncertain environment.

The development of a bridge between the activity choice behaviour and the longstanding network equilibrium models has attracted considerable research efforts. The supernetwork representation is adopted as a unified framework for modelling activity location, time of participation, duration, and route choice in congested networks. However, imposing simple constraints on the activity-travel schedules makes the supernetwork models computationally intractable. In this thesis, the supernetwork models are shown to be special cases of the MDP model. The computational burden can be alleviated by exploiting the special structure of the MDP model. Dynamic programming algorithms are developed to solve the MDP model without enumeration of all the possible activity-travel schedules.

In traditional travel demand models, travellers within a household are treated separately. Each household member makes activity-travel decisions independently. As a result, the estimation of joint activity participation can be biased. In this thesis, a household MDP model is proposed to explore the influence of intra-household interactions on individual's activity-travel scheduling behaviour. A utility function is adopted to represent joint household preference. The function consists of a weighted sum of each household member's utility, together with a term measuring the level of intra-household interactions.

The variation in intra-household interactions across activity types is thoroughly examined using the household MDP model. The intra-household interactions are considered as weak or negligible for compulsory activities, such as work and school. These activities are conducted with significant regularity and under strict spatial and temporal constraints. Joint non-compulsory activities are motivated by collaboration and companionship. The intra-household interactions are positive for these activities. For example, one household member's choice of activities, such as social visits and outdoor sports, is highly related to the other's choice. The intrahousehold interactions are negative for other non-compulsory activities, such as cooking and cleaning. If one household member has completed one such activity, the other does not have to undertake this activity but still benefits from its results. To minimize the effort spent on these activities, a household simply allocates each of these activities to one household member.

## **Publications Arising From the Thesis**

The following papers were published during the author's MPhil study period:

- Yiliang Xiong, William H.K. Lam. An Activity-Based Approach for Estimation of Passenger O-D Matrix and Trip Chain Pattern. World Conference on Transport Research 2010, Lisbon, Portugal.
- Yiliang Xiong, William H.K. Lam. Modelling Within-day Dynamics in Activity Scheduling: A Markov Decision Process Approach. Journal of the Eastern Asia Society for Transportation Studies, Vol. 9 (2011), 452–467.

#### Acknowledgments

I want to thank my chief supervisor, Chair Professor William H.K. Lam, for sharing his knowledge on transportation research, and for his patience with my twisty way of learning and doing research. This thesis could never reach the current state without his help and insightful comments.

I would like to thank my co-supervisor Dr. Agachai Sumalee, for being interested in my research, and giving me tremendous suggestions. I also extend my sincere gratitude to Professor Becky P.Y. Loo and Professor Donggen Wang, who served as external examiners on the examination board and gave valuable comments on this thesis.

Finally, I want to thank my family and friends for their understanding and support on this journey.

## **Table of Contents**

Abstracti
Publications Arising From the Thesisiv
Acknowledgmentsv
Table of Contentsvi
List of Figuresxi
List of Tablesxiii
List of Notationsxiv
Chapter 1 Introduction
1.1 General background1
1.2 Need for the study
1.3 Research objectives
1.4 Structure of the thesis
Chapter 2 Literature Review7
2.1 Introduction7
2.2 Evolution of the travel demand modelling approaches7
2.2.1 Activity-based network equilibrium models10
2.2.2 Activity-based supernetwork equilibrium models
2.3 Utility of activity-travel choice
2.3.1 Process utility and goal utility
2.3.2 Utility of the inter-temporal choice

2.4 Dynamics in activity-travel behaviour	14
2.4.1 Within-day dynamics	15
2.4.2 Day-to-day dynamics	15
2.4.3 Sequential decision process	16
2.4.4 Markov decision process	17
2.5 Intra-household interactions	18
2.6 Summary	19
Chapter 3 Study Methodology	20
3.1 Introduction	20
3.2 Markov decision process	20
3.2.1 Modelling framework	21
3.2.2 Markov property	23
3.2.3 Solution algorithms	24
3.3 An introductory example	25
3.3.1 Enumeration approach	26
3.3.2 Recursive method	27
3.3.3 Random errors	28
3.4 Summary	30
Chapter 4 Within-Day Dynamics in Individual's Activity-Travel Sector	cheduling
Behaviour 32	
4.1 Introduction	32
4.2 An individual's activity-travel scheduling model	33

4.2.1	Assumptions
4.2.2	Model formulation
4.2.3	Marginal utility
4.2.4	Decision set
4.2.5	Deterministic state transition
4.2.6	Probabilistic state transition
4.2.7	Random errors
4.2.8	Relationship with activity-based supernetwork equilibrium models
	46
4.3 Solu	ution algorithms
4.4 Nur	nerical examples
4.4.1	Discount ratio for future utility
4.4.2	Value of travel time
4.4.3	Effect of increasing travel time
4.5 Sun	nmary
Chapter 5 In	ntra-Household Interactions in Household's Activity-Travel
Scheduling Bo	ehaviour
5.1 Intro	oduction
5.2 An	activity-travel scheduling model with intra-household interactions .60
5.2.1	Assumptions
5.2.2	Model formulation
5.2.3	Household utility function

5.2.4	Compulsory and non-compulsory activities	67
5.3 Solu	ition algorithms	68
5.4 Nun	nerical examples	70
5.4.1	Positive intra-household interaction for shopping	73
5.4.2	Negative intra-household interaction for shopping	76
5.5 Sum	nmary	77
Chapter 6 C	alibration Methods and Results	79
6.1 Intro	oduction	79
6.2 Max	kimum likelihood method	79
6.3 Nun	nerical methods for parameter calibration	80
6.3.1	Nested fixed-point algorithm	80
6.3.2	Mathematical programming with equilibrium constraints	81
6.4 Cali	brating the individual's MDP model	81
6.4.1	Marginal utility functions	
6.4.2	Activity-travel data generation	
6.4.3	Calibration results	85
6.5 Cali	brating the household's MDP model	
6.5.1	Household activity-travel data generation	
6.5.2	Calibration results	90
6.6 Sum	nmary	93
Chapter 7 C	onclusions and Discussion	94

7.1	Summary and conclusions	94
7.2	Limitations of the study and future research	96
7.2.	1 Challenges in data collection	98
Appendix	x I: MDP with temporal abstraction	101
Appendix	x II: AMPL code for solving MDP models and calibrating 1	nodel
paramete	ers	105
Referenc	ces	110

## List of Figures

Figure 1.1 Temporal profile of the marginal activity utility
Figure 1.2 The household's daily activity-travel scheduling
Figure 1.3 The interdependencies among chapters
Figure 3.1 Value iteration method for standard MDP model25
Figure 3.2 A 2-node transportation network
Figure 3.3 The overall utility for all possible departure times
Figure 3.4 Expected maximum future utility over time of the day
Figure 3.5 Activity participation of the entire day
Figure 4.1 Marginal activity utilities of three activities over 24 hours
Figure 4.2 Solution algorithms for the MDP model and the network equilibrium
problem
Figure 4.3 Algorithm for calculating variables f(s)
Figure 4.4 Algorithm for calculating flow variables
Figure 4.5 A 3-node network
Figure 4.6 Temporal profiles of marginal utility functions
Figure 4.7 Activity participation over time of the day
Figure 4.8 Activity participation over time of a typical traveller ( $\beta = 0.99$ and $\beta =$
0.97)
Figure 4.9 Activity participation of a typical traveller ( $\alpha = 60$ and $\alpha = 120$ )
Figure 4.10 Activity participation of a typical traveller under different traffic
conditions

Figure 5.1 The utility of joint activity participation for the entire household
Figure 5.2 Dynamic merging algorithm for household MDP model
Figure 5.3 A 4-node road network
Figure 5.4 The individual's utility for independent activity participation with
heterogeneous preferences
Figure 5.5 Activity participation of a two-person household ( $\rho = 0$ )
Figure 5.6 The utility of joint activity participation
Figure 5.7 Activity participation of a two-person household ( $\rho = 0.2$ )
Figure 5.8 The utility of independent activity participation with homogenous
preferences
Figure 5.9 Activity participation over time of the day ( $\rho = -0.2$ )
Figure 6.1 Temporal profiles of two types of marginal utility function
Figure 6.2 A 3-node transportation network
Figure 6.3 Temporal profile of the marginal utility function
Figure 6.4 The log-likelihood function $l_i(b_2)$
Figure 6.5 Contour and 3-D plot of the log-likelihood function $l_i(b_2, b_3)$
Figure 6.6 A 4-node road network
Figure 6.7 Temporal profile of the marginal utility function
Figure 6.8 The log-likelihood function $l_i(\rho_3)$
Figure 6.9 Contour and 3-D plot of the log-likelihood function $l_i(\rho_1, \rho_3)$ 92
Figure 7.1 GPS points, trajectories and an activity location

## List of Tables

Table 2.1 Comparison of travel demand modelling approaches    8
Table 3.1 Marginal activity utility at each time episode    26
Table 4.1 Temporal constraints and parameters of marginal utility function51
Table 4.2 Allocation of time to activities and travel under different traffic
conditions
Table 5.1 The utilities of independent activity participation
Table 5.2 Parameters of utility function for each activity71
Table 5.3 Allocation of time to activities and travel for different values of $\boldsymbol{\rho}$
(Individual 2)
Table 6.1 Parameters of marginal utility function    84
Table 6.2 Calibration results of parameters in the marginal utility function
Table 6.3 Parameters of marginal utility functions for the household
Table 6.4 Calibration results of parameters in the marginal utility function for each
household member

## List of Notations

Notations	Notes
S	The set of states
D	The set of decisions
С	The set of primitive choices
β	The discount ratio of the future utility
p(s' s,c)	The state transition probability function in standard MDP
p(s' s,d)	The state transition probability function in MDP with temporal abstraction
r(s,c)	The deterministic component of the immediate utility $R$ in standard MDP
r(s,d)	The deterministic component of the immediate utility $R$ in MDP with temporal abstraction
$\mathcal{E}(s,c)$	The random component of the immediate utility $R$ in standard MDP
$\varepsilon(s,d)$	The random component of the immediate utility $R$ in MDP with temporal abstraction
$G_{\!arepsilon}\left(\cdot ight)$	The cumulative distribution function of $\varepsilon$
$\pi(s,c)$	A mapping from states to the probabilities of taking each choice in standard MDP
$\pi(s,d)$	A mapping from states to the probabilities of taking each decision in MDP with temporal abstraction
V(s)	The expected discounted utility at any state $s$
Q(s,c)	The expected utility of selecting any choice $c \in C(s)$ at state <i>s</i> in standard MDP
Q(s,d)	The expected utility of selecting any decision $d \in D(s)$ at state

#### *s* in MDP with temporal abstraction

Ν	The set of activity programs, $\{1, 2, N\}$
$A^n$	The set of activities in activity program $n$
Т	The planning horizon, i.e., number of time episodes
t <sub>s</sub>	The time component of state <i>s</i> , $t_s \in \{1,, T\}$
w <sub>s</sub>	The location of the traveler at state $s$
$a_s$	The on-going activity at state $s$
$e_s$	The remaining time of the on-going at state $s$
$\underline{t}_a, \overline{t}_a$	The earliest and latest time for activity $a$
W	The set of available activity destinations
B(w)	The set of available activities at location $w$
$A_{s}$	The set of the uncompleted activities at state $s$
$\overline{A}_s$	The set of the feasible activities that can be chosen at state $s$
$a_d$	The activity type component of decision $d$
$h_d$	The activity duration component of decision $d$
Z <sub>d</sub>	The activity location component of decision $d$
<i>m</i> <sub>d</sub>	The travel mode component of decision $d$
V.	A route in road network, denoted by $y_{d,r}$ if it is a component of
Jr	decision d
$y_b$	A bus line in transit network, denoted by $y_{d,b}$ if it is a
	component of decision $d$
$\tau(t,w,z,m)$	The time needed to travel from $w$ to $z$ departing at time $t$
$\alpha_{_m}$	The equivalent disutility of unit travel time for travel mode $m$

u(a,t)	The discrete marginal utility of activity $a$ at time $t$
$g_a(t)$	The continuous marginal utility of activity $a$ at time $t$
$\sigma_{i}$	The weight parameter representing the relative influence of household member $i$
r <sub>i</sub>	The utility of independent activity participation for household member $i$
$r_{J}$	The utility of joint activity participation for a household
ρ	The level of interaction between household members
$G_s$	The set of non-compulsory activities that can be taken by any household member at state $s$
I	The set of states in which the on-going activity or travelling is completed, $\mathcal{I} \subset S$
f(s)	The number of traveller in state <i>s</i>
$f_l(t)$	The number of travellers choosing road link $l$
θ	The vector of model parameters to be calibrated
$l_i( heta)$	The contribution of individual $i$ to the log-likelihood function
S <sub>it</sub>	The observed state of individual $i$ at time $t$
$d_{_{it}}$	The observed decision of individual $i$ at time $t$

## **Chapter 1 Introduction**

#### 1.1 General background

The activity-based travel demand modelling systems commonly have hierarchical structures (Bowman and Ben-Akiva, 2001). These structures can be divided into two levels. The upper level involves activity generation and produces a set of activity programs. The lower level focuses on activity-travel scheduling behaviour, involving choices of activity destination, timing and duration.

The hierarchical structure of activity-based simulation models is analogous to the sequential steps in the four-step model, namely trip generation, trip distribution, travel mode choice and route choice. The simulation models generate activities, tours, and trips with flexible and complex cross-impacts on one another.

This thesis is motivated by the widely accepted concept that travel demand is derived from the need of participating in spatially separated activities. The travel demand is therefore treated as an outcome of activity choice. The goal of activity scheduling is to maximize the overall utility of engaging in activities. The activity utility varies over time of the day. For a given activity program, the sequence of activity participation is affected by the temporal profile of marginal activity utility.

Figure 1.1 illustrates the marginal utility of three activities for a normal weekday (24 hours). The optimal activity sequence is Home  $\rightarrow$  Work  $\rightarrow$  Shopping  $\rightarrow$  Home. The optimal switching time between activities reflects the temporal profile of marginal activity utility. The departure time of each trip is then determined.



Figure 1.1 Temporal profile of the marginal activity utility

The second motivation is to develop a framework for modelling intra-household interactions associated with distinct activity types. The daily activities can be categorized into two types based on the flexibility of participation, the compulsory and non-compulsory activities. Firstly, the individuals' choices of compulsory activities, such as work and school, are independent from other household members. These activities are undertaken with significant regularity (every weekday) and under strict spatial and temporal constraints.

Secondly, non-compulsory activities, such as cooking, cleaning, grocery shopping and pick-up children from school, are usually allocated to one household member. Typically, a household is only willing to spend a minimum amount of time or the least effort to complete such activities. For example, either the husband or the wife is responsible for sending children to school.

Thirdly, collaboration and companionship motivate joint activity participation, such as social visits and having dinner outside with family. Joint participation is more efficient and satisfactory for household members than independent participation. Figure 1.2 shows that in the morning the husband escorts his wife to her workplace and then heads for his own workplace. In the evening, the husband has dinner with his wife.



Figure 1.2 The household's daily activity-travel scheduling

#### **1.2** Need for the study

The current development of the activity-travel scheduling models lacks a rigid and comprehensive modelling framework. Most current activity-based network equilibrium models cover only a few choice dimensions. The specification of choice dimensions is based either on the available travel survey data or on a relatively ad hoc method. Hence, this study sets out to develop a framework for modelling individual and household activity-travel scheduling behaviour in congested transportation network.

Daily activity-travel scheduling is a dynamic decision process. The dynamics in activity-travel scheduling have a two-fold meaning. First, travellers make activity-travel decisions repeatedly over time. The contextual situations at the decision epoch, such as time of the day and the traveller's location, play a crucial role in the

decision-making process. Second, a rational traveller anticipates and evaluates the possible future consequences when making decision at the current moment. The traveller's present decision influences the future situations and thus has an impact on the future decisions.

It was recognised in the activity-based simulation models that the interaction between household members would influence one another's activity choices particularly in a congested transportation network. Certain types of activity can be assigned to a particular household member. Other household members would benefit from that household member's action. Household members also jointly participate in activities to obtain higher overall utility for the entire household.

The household's activity-travel scheduling behaviour is challenging for modelling framework development. A main challenge is the considerable variation in intrahousehold interactions across activity types. The intra-household interactions are negligible for compulsory activities, such as work and school, but play a central role in non-compulsory activities, such as social visits and outdoor sports. Another challenge is that the number of decisions available to a household with several members is greater than that of an individual. The larger decision space imposes a marked computational burden on the solution process.

#### **1.3 Research objectives**

The aim of this thesis is to incorporate within-day dynamics and intra-household interactions into the existing activity-based network equilibrium modelling methodologies. The main objectives are as follows.

 To propose a unified and extensible activity-based network equilibrium modelling framework for activity-travel scheduling. This framework covers multi-dimensional choices and allows constraints imposed on these choices.

- 2. To capture the dependency of the traveller's choice on the contextual situations, and investigate how the expectation of the future affects the present decision.
- To model and quantify the impact of intra-household interactions on individual household member's activity-travel decision.
- To develop methods for calibrating model parameters, such as parameters of marginal activity utility function and the coefficients of intra-household interactions.

#### **1.4 Structure of the thesis**

The background and the motivations of this thesis have been described in Chapter 1. The remainder of this thesis is organized as follows. The interdependencies among chapters are illustrated in Figure 1.3.

Chapter 2 gives a review on literature on travel demand modelling. The evolution of travel demand modelling is briefly reviewed. The review then focuses on the relevant topics in activity-travel scheduling behaviour.

Chapter 3 describes the study methodology and preliminary knowledge that are required to understand the modelling methodology adopted in this thesis. An introductory example is presented to give some intuitions of the methodology.

Chapter 4 explores the with-day dynamics in individual's daily activity scheduling behaviour. Markov Decision Process (MDP) is employed as a unified framework for formulating individual's scheduling behaviour in a congested network. The supernetwork models are shown to be special cases of the MDP model.

Chapter 5 presents an MDP model with intra-household interactions, which is a direct extension of the individual's activity-travel scheduling model proposed in

Chapter 4. The impact of the intra-household interactions is analysed in a few numerical examples.

Chapter 6 introduces calibration methods for the individual and household activitytravel scheduling models. Numerical methods for solving the model calibration problem are optimized based on the mathematical property of the models. Hypothetical numerical experiments are conducted to generate time-series data for model calibration.

The final chapter summarises the findings and limitations of this thesis and gives directions for future research.



Figure 1.3 The interdependencies among chapters

#### **Chapter 2** Literature Review

#### 2.1 Introduction

A brief review on the evolution of the travel demand modelling approaches is presented at the beginning of this chapter. The review focuses on network equilibrium models for activity-travel choices. Since utility maximization theory is the standard framework for evaluating activity-travel choices, the concept of activity-travel utility is also reviewed.

The findings from empirical studies on dynamics in activity-travel scheduling are summarized in support of the dynamic behaviour studied in this thesis. The activity-travel scheduling behaviour can be viewed as a sequential decision process. Previous transportation studies on sequential decision process can be considered as prototypes of Markov Decision Process (MDP) models. The advantages of MDP models are explained in the context of activity-travel scheduling behaviour. Finally, previous studies on household intra-household interactions are reviewed to support the methodologies adopted in this thesis.

#### 2.2 Evolution of the travel demand modelling approaches

The fundamental objective of transportation planning is to provide appropriate transportation facilities to meet the future travel demand (Lam et al., 2006; Li et al., 2008; Oppenheim, 1995). The knowledge of travel demand can help improve the efficiency and sustainability of the existing transportation systems (Hatzopoulou et al., 2007; Sheffi, 1985). Travel demand forecasting is a long-standing topic in transportation research. Various approaches have been proposed to modelling travel demand. Table 2.1 presents the features of three types of travel

demand models. The evolution of travel demand modelling approaches is briefly reviewed in the remainder of this section.

	Trip-based approach	Trip-chaining approach	Activity approach
Unit of analysis	Trip	Trip chain or Tour	Activity
Decision unit	Traffic analysis zone	Individual traveller	Household
Model structure	Sequential	Combined	Hierarchical
Solution algorithm	Numerical method	Heuristic method	Simulation*
Computational burden	Low	Intermediate	High

Table 2.1 Comparison of travel demand modelling approaches

\*: Most existing activity-travel scheduling models are based on simulation method, except for the activity-based network equilibrium models reviewed in Sections 2.2.1 and 2.2.2.

In late 1980s, the research efforts have been condensed into a complete and consistent modelling framework, the four-step method, shown as traditional approach in Table 2.1 (Oppenheim, 1995; Sheffi, 1985). The four-step method adopts *trip* as the basic unit of analysis and includes four sequential steps: trip generation, trip distribution, mode split and traffic assignment. The studied area is divided into several traffic analysis zones. In trip generation step, the production and attraction of trips in each zone are derived from the socio-economic characteristics of the zone. In trip distribution step, the number of trips between each pair of zones is computed. Trips between zones are required to be consistent with the production and attraction of each zone. In mode split step, discrete choice model is employed to determine the proportion of trips that are implemented by a specific travel mode (e.g., private car, bus). Finally, in traffic assignment step, these trips are assigned onto road links and transit routes in the transportation system. The travellers are assumed to choose the shortest or the minimum cost route.

Transportation network plays a central role in network equilibrium models. Travellers choose trip destination, departure time, travel mode and route based on the perception of travel cost in congested networks. These models are formulated as equivalent mathematical programming problems and solved with numerical methods (Florian and Hearn, 1995).

The trip-chaining model, shown as transitional approach in Table 2.1, is a direct extension of the four-step method. A trip chain is a set of ordered trips connecting consecutive stops, for example, Work  $\rightarrow$  Shop  $\rightarrow$  Home. The trip-chaining model takes account of the relationship between individual trips. Trips in the same trip chain influence one another in terms of total travel time and mode choice. For example, if a traveller drives to work in the morning, the traveller has to drive the car back home in the evening. The mode choice in the morning affects the choice in the evening. Since the emergence of activity-based simulation models, trip chain has been also adopted as a connection between activities and trips.

Abdelghany et al. (2001; 2003) implemented a system to simulate the traveller's choice of trip chains in a real-world transportation network. A user equilibrium solution was obtained through iterative simulation-assignment procedure. Maruyama and Harata (2005; 2007) formulated the trip-chaining behaviour as a convex nonlinear programming problem. Their trip chain choice model was built on the user equilibrium principle.

The activity-based simulation models, shown as existing approach in Table 2.1, for analysing travel behaviour and predicting travel demand emerged in the late 1980s (Jones, 1990; Kitamura, 1988; McNally, 2000). These simulation models focus on predicting where and when activities are pursued. The daily activities play a central role. Travel is considered as a demand derived from the need for participating in spatially separated activities. In the traditional four-step model, travel is a desirable activity on its own right. The methodologies in the previous activity-based simulation models rely on a particular dataset or one perspective of travel behaviour rather than a unified modelling framework. The aspects of activity-travel behaviour modelling that have been examined are interesting but limited, such as the hierarchical structure of activity-travel choice (Bowman and Ben-Akiva, 2001), development of micro-simulation models (Abdelghany et al., 2001; Abdelghany and Mahmassani, 2003; Kitamura et al., 1996), applications of econometric theory (Bhat, 1996; Golob, 2000).

The network structure is ignored in many activity-based simulation models. Travel time between activity locations is treated as fixed and known. Travel demand is purely derived from activity participation. These models are commonly built on econometric theory (Bhat, 1996; Bowman and Ben-Akiva, 2001; Golob, 2000) or micro-simulation method (Arentze et al., 2000; Garling et al., 1994; Kitamura et al., 1996). Complicated activity-travel behaviour can be modelled in activity-based simulation models (Arentze et al., 2000; Garling et al., 1994; Kitamura et al., 1996). Roorda et al., 2008). However, most simulation models rely on ad hoc rules and lack a rigorous theoretical foundation.

#### 2.2.1 Activity-based network equilibrium models

The activity-based simulation models and network equilibrium models have been independently developed for years. Separate groups of researchers investigate the travel demand forecasting problem from these two different perspectives (Kitamura, 1988; Mahmassani, 1988). Most activity-based simulation models lack a rigorous theoretical foundation and simulation must be repeated many times to obtain a stable solution. However, network equilibrium models have analytical forms and their solutions can be computed by numerical methods.

10

Considerable research efforts have been made on the development of a bridge between the activity-travel choice behaviour and the long-standing network equilibrium problem. Lam and Yin (2001) combined activity choice and the timedependent network equilibrium model. The travellers are assumed to care only about the immediately utility obtained in a short period. The activity with the highest immediate utility is chosen in each period. Lam and Huang (2003; 2005) extended the combined model to dynamic and stochastic user equilibrium.

Lin et al. (2008) developed a conceptual framework by combining activity choice and dynamic network equilibrium. The network equilibrium model is used to generate dynamic travel time in their model. The travellers schedule daily activities with the knowledge of dynamic traffic conditions. Following this line of research, recent research efforts have been directed towards the development of a unified modelling framework.

#### 2.2.2 Activity-based supernetwork equilibrium models

The supernetwork representation is a common framework for modelling joint travel choice among network equilibrium models, such as the joint mode split/distribution/assignment model (Sheffi, 1985). Recently, this framework is adapted to model dynamic activity-travel choices. The physical transportation networks are augmented with virtual nodes and links to represent activity-travel choices.

Ramadurai and Ukkusuri (2010, 2011) proposed a dynamic network equilibrium model to capture activity location, time of participation, duration, and route choice jointly. Virtual links are added to represent additional choice dimensions. The activity-travel schedules are represented by routes on the extended network.

Ouyang et al. (2010, 2011) defined the Activity-Time-Space (ATS) network by augmenting a physical transportation network. The nodes in the expanded network

denote all the possible time and space choices for a traveller. There are two types of links in the expanded network. The activity link indicates the process of activity participation for one time episode. The road link represents travel between physical nodes. Each link is associated with a utility or disutility function, which specifies the preference of the traveller. The equilibrium flows are obtained by applying stochastic network equilibrium algorithms on the expanded network.

Each route in the ATS network is supposed to associate with an activity-travel schedule and thus, only the routes that satisfy all the conditions imposed on schedules are feasible. For example, it is reasonable to require that every daily activity-travel schedule begins with in-home activity and ends with in-home activity. There is however, no neat solution to the additional constraints in the framework of supernetwork representation.

#### 2.3 Utility of activity-travel choice

Utility maximization theory is the standard framework for measuring the benefit and loss in most activity-based simulation models. An individual's activity-travel choice is viewed as a reflection of underlying utilities associated with each of the choice alternatives. The individual selects the choice alternative providing the highest utility.

#### 2.3.1 Process utility and goal utility

The concepts of process utility and goal utility are proposed to express different form of benefits in activity participation (Axhausen and Garling, 1992; Winston, 1987). The process utility, which measures the utility that continuously derived during the activity participation, is related to activity timing and duration (see Figure 1.1). The goal utility, which represents the utility obtained by finishing the activity, is related to the consumption behaviour and/or the objectives achieved in the activity. The total utility of engaging in a certain activity is the sum of these two types of utilities.

Ettema and Timmermans (2003; 2004) proposed a temporal profile function for process utility of activity participation, which depends on activity timing and duration. It is suitable for activities like working and studying, where people spend time and derive utility in the process of conducting activities. However, for other activities, such as shopping and eating outside, the amount of good consumption during activity participation also affects the activity utility. Additionally, the marginal utility should diminish with the amount of consumption.

On one hand, modelling activity program formation should be based on goal utility measurement. On the other hand, the research of activity-travel scheduling focuses on the process of activity participation. Activity timing and duration plays a central role in activity-travel scheduling. Thus, the process utility is adopted in this thesis to represent the traveller's preference of activity-travel schedules. The overall utility of an activity-travel schedule is the sum of activity utility and disutility derived from travelling between activity destinations.

#### 2.3.2 Utility of the inter-temporal choice

Instead of looking merely at the total time allocated to an activity, Winston (1982, 1987) investigated individuals' inter-temporal choice of consumption and time use. The activity timing and duration choices reflect the marginal utility of activity participation, which varies over time of the day (Supernak, 1992).

In this thesis, the utility of inter-temporal activity-travel choice is based on the discounted utility model (Charypar and Nagel, 2005). The value of a sequence of inter-temporal choices  $d = (d_1, ..., d_T)$  is expressed by the weighted utility formula:

$$V(d) = \sum_{t} w_t \cdot u(d_t)$$
(2.1)

Equation (2.1) implies that whenever two sequences of choices differ in only two periods, the preference over them does not depend on the choices in other periods. Two additional assumptions are widely employed to evaluate inter-temporal choice:

(1) The weights,  $w_1$ ,  $w_2$ , and so forth, are declining, which indicates the individual values the utility derived in earlier periods more than that in later periods.

(2) The marginal rate of discounting between any two consecutive periods is the same  $w_{t+1}/w_t = \beta$ . This assumption leads to the discounted utility with constant-rate,  $V(d) = \sum_t \beta^t \cdot u(d_t)$ .

#### 2.4 Dynamics in activity-travel behaviour

In recent years, the dynamics in activity-travel behaviour attract considerable research effort. Activity-travel choices are found to have strong dependency over time and history of activity participation. Two types of dynamics in activity-travel behaviour are presented in Section 2.4.1 and 2.4.2. Most of these studies are based on simulation method, but their findings can be used to support the activity-travel scheduling behaviour studied in this thesis.

Most repeated activity-travel choices over time can be viewed as a sequential decision process. The idea of sequential decisions has been found in transportation research for a long time. In this thesis, the activity-travel scheduling behaviour is modelled as a Markov decision process. The modelling methodologies for repeated activity-travel choices over time are reviewed in Section 2.4.3 and 2.4.4.

#### 2.4.1 Within-day dynamics

One aspect of the within-day dynamics in activity-travel scheduling is that most activities are pursued once in a day, and thus, the current choice of activity depends on the history of activity participation (Kasturirangan et al., 2002). Generally, the choice of activity at present is closely related to the state of the traveller, such as the remaining time for out-of-home activities and the activities pursued before the choice.

Habib (2010) proposed an econometric modelling framework for daily activity program generation. Composite activity is introduced to integrate all the activities in the rest of the day. Travellers balance the time allocated to a specific activity and the composite activity in order to maximize the overall utility that can be obtained in the entire day.

#### 2.4.2 Day-to-day dynamics

Cirillo and Axhausen (2010) formulated activity-travel choices in different days as a discrete choice model with correlation between inter-temporal choices. The present activity utility is assumed to depend on the past choice of activity type and the number of days passed since last activity participation. For example, the weekly grocery shopping is conducted once a week. The purchased food and other products are consumed in the next a few days. Thus, the probability of making another shopping trip increases with the number of days passed since the last grocery shopping.

A need-based model of multi-day, multi-person activity generation is proposed in Arentze and Timmermans (2009). The concept of need represents the amount of daily goods or the emotional desire for social activity. The latent attributes, such as the individual's need and time pressure, influence the activity utility. The activity utility is assumed to vary over time since the need for the activity differs over time. Accordingly, the utility of pursuing an activity is a function of the individual's needs. Needs grow over time and when a threshold is reached, an activity is selected to satisfy the needs. Pursuing an activity can satisfy several needs, for example, having dinner with a friend fulfils the needs of food and social gathering.

#### 2.4.3 Sequential decision process

In transportation literature, the idea of sequential decisions was first proposed by Dial (1971). In Dial's paper, the route choice problem was considered as a sequence of choices of links. The Dial's network loading algorithm was proved equivalent to a logit route choice model, which was criticized for the restraints imposed by the IIA property. In addition, the algorithm only applies to a subset of the routes that are composed of the predefined efficient links.

Kitamura (1984) proposed "prospective utility" as the measurement of attractiveness of traffic zones. The attractiveness of a zone is not only determined by the accessibility and attributes of the zone but also by other opportunities that can be reached from the zone. For example, if a traveller drives to a shopping mall in the downtown, the traveller takes into account the utilities that can be obtained from adjacent activity centres. The prospective utility of a zone is thus defined as the weighted sum of the utilities that can be obtained from the zone and from potential future trips.

Baillon & Cominetti (2008) employed dynamic programming to solve the route choice problem. The traveller chooses a link to follow at every intermediate node on the trip to the destination. Route choice is the outcome of the sequential decision process. Distinct discrete choice models can be adopted at every node. User equilibrium and stochastic user equilibrium are two special cases that fit into this unified modelling framework.
#### 2.4.4 Markov decision process

Traditionally, analytical models in transportation literature are formulated in the form of nonlinear programming and variational inequality problems (Dafermos, 1980; Sheffi, 1985). These theories provide general frameworks for formulating and solving various network equilibrium models. Since this thesis focuses on the within-day dynamics in activity-travel scheduling, Markov Decision Process (MDP) is a more expressive framework for formulating the complicated intertemporal choices in activity-travel scheduling (Puterman, 1994). Furthermore, MDP can be integrated with network equilibrium models without of difficulty. The activity-based supernetwork equilibrium models are shown to be special cases of the MDP model for activity-travel scheduling in Section 4.2.8.

The advantage of this approach is that it does not need to consider each activitytravel schedule individually. There is thus no need to enumerate all the possible schedules. Another feature of the MDP formulation is that it takes into account the expected utility that can be obtained in the near future. The traveller makes optimal decision at present with the knowledge that he will also make optimal decisions in the future. Usually, this behaviour is hard to formulate in the other modelling frameworks.

Comprehensive reviews on MDP models can be found in (Aguirregabiria and Mira, 2010; Eckstein and Wolpin, 1989; Rust, 1994). The structural dynamic discrete choice models are built upon the framework of the discrete Markov decision processes proposed by Rust (1994). Previous applications of MDP in activity scheduling behaviour can be found in Jonsson and Karlström (2005). Charypar and Nagel (2005) applied a similar model from computer science. Recently, Cirillo and Xu (2011) presented an extensive review of MDP models from the perspective of transportation research.

## 2.5 Intra-household interactions

Discrete choice models are widely employed in the activity-based travel demand modelling literature. An important assumption usually adopted in discrete choice models is that each individual's choice is independent of that of other individuals (Ben-Akiva and Bierlaire, 1999; Train, 2003). This assumption is, however not satisfied in the context of household activity-travel scheduling. The interdependencies between household members influence the activity participation of each household member.

Even though the household consists of different individuals, the household is commonly assumed to act as a single decision-making unit in this thesis. Hence, the household activity choice fits into the modelling framework of discrete choice. A household utility function can be proposed to capture the joint household preference with consideration of the intra-household interactions (Bradley and Vovsha, 2004; Zhang et al., 2009). The household utility function includes the weighted sum of each household member's utility. Additional terms are incorporated to capture the intra-household interactions. A general form of the household utility function can be expressed as:

$$U(v) = v'Av + b'v \tag{2.2}$$

where  $v = (v_1, v_2, ...)'$  is the vector of individual's activity utility. *A* is a symmetric matrix with entries  $\rho_{ij}(i, j = 1, 2, ...)$  which measures the level of intrahousehold interactions. Weight parameter  $b = (\sigma_1, \sigma_2, ...)'$  represents the relative influence of household member.

Miller and Roorda (2003) and Roorda et al. (2008) presented comprehensive simulation models for household activity-travel scheduling. A synthetic population is used to represent the households for an urban region. Detailed activity-travel

schedules over twenty-four hours are generated for all individuals in a household in the simulation. Activity frequency, starting time and duration are drawn from probability distributions. The activity episodes are scheduled based on predefined rules. Intra-household interactions are captured through the generation of joint activity episodes in which more than one person in the household participate in.

To capture the conflicts and bargaining within household, game theory is widely adopted in microeconomic literature (McElroy and Horney, 1981). A household is considered as a group of individuals with different preferences, among whom a bargaining process takes place to resolve the conflicts on activity-travel choices. In the cooperative framework, household members come to an agreement on allocation of welfare within household. Although game theory can be used to capture a rich range of activity-travel behaviour, modelling the conflicts and bargaining among household members are out of the scope of this thesis.

## 2.6 Summary

The literature related to the methodology and basic concepts in this thesis are reviewed in this chapter. The evolution of activity-based travel demand modelling and the main approaches for analysing activity-travel behaviour are reviewed in Section 2.2. Then the concept of activity-travel utility is introduced in Section 2.3. The literature on dynamics in activity-travel behaviour and intra-household interactions are described in Sections 2.4 and 2.5. These two sections present the background of the individual's and household's activity-travel scheduling models proposed in Chapter 4 and 5.

## Chapter 3 Study Methodology

## **3.1 Introduction**

The basics of the adopted mathematical modelling framework are presented in this chapter. Markov Decision Process (MDP) for modelling dynamic behaviours is described in Section 3.2. An introductory example on departure time choice is presented in Section 3.3. This example illustrates the application of MDP in travel behaviour modelling.

## 3.2 Markov decision process

Nonlinear programming provides a general framework for formulating and solving optimization problems. Similarly, MDP provides a structure within which optimal control of dynamic systems can be formulated and solved (Puterman, 1994). MDP is extensively documented in economic and marketing literature (Aguirregabiria and Mira, 2010; Eckstein and Wolpin, 1989). In this thesis, the MDP model is employed to investigate the dynamic behaviour in activity-travel scheduling.

Travellers make activity-travel choices repeatedly over time. The choices depend on the contextual situations, such as time of the day and the traveller's location. The contextual situations can be represented by state in the MDP model. A rational traveller attempts to anticipate the future situations and gain more enjoyment out of the day. This behaviour is captured by the objective of the MDP model. With appropriate definition of state, state transition and objective function, the activitytravel scheduling behaviour is grounded in a rigorous mathematical framework and viewed from a broader perspective.

#### 3.2.1 Modelling framework

MDP models can be categorized into two main groups based on how the time is modelled: the discrete-time MDP and the continuous-time MDP. In discrete-time MDP model, the planning horizon is divided into equal periods. It is reasonable to assume that the traveller takes activity-travel choices at discrete decision epochs and thus, discrete-time MDP model is adopted in this thesis.

A discrete-time MDP model consists of the following components:

- (1) A time episode used to index the decision process k = 1, 2, 3, ...
- (2) A set of states  $s_k \in S$ .
- (3) A set of primitive choices  $c_k \in C(s_k)$ , let  $C = \bigcup_{s \in S} C(s)$  denote the union of the choice sets.
- (4) A state transition probability function  $p: S \times S \times C \rightarrow [0,1]$ .
- (5) An immediate utility function  $r(s_k, c_k)$ .
- (6) A discount ratio for future utility  $\beta$ .

The MDP model describes the behaviour of a traveller who makes choices with multiple components repeatedly over time. The time episode k is used to index the choices. At any time episode k, the traveller perceives the contextual situations that is represented by state  $s_k$  and selects a primitive choice  $c_k \in C(s_k)$ . The choice dimensions include the activity to be pursued, the location of the activity, etc. After making a choice, the traveller implements it immediately. The traveller receives the immediate the choice utility derived from  $r(s,c) = E[r_k | s_k = s, c_k = c]$ . Then the state of the traveller advances to  $s_{k+1}$  in the next time episode k+1.

As the state transition depends on (s, c), the subsequent state is thus written as s' | s, c to indicate this dependency. The state transition is modelled by the

probability function  $p(s'|s,c) = \Pr[s_{k+1} = s'|s_k = s, c_k = c]$ , where the sum of probabilities over all states is one  $\sum_{s' \in S} p(s'|s,c) = 1.0$ .

A *Markov policy* is a mapping from states to probabilities of taking each primitive choice  $\pi: S \times C \rightarrow [0,1]$ . The traveller's objective is to find a Markov policy that maximizes the expected discounted utility at any state *s*:

$$V^{\pi}(s) = E \Big[ r_{k} + \beta r_{k+1} + \beta^{2} r_{k+2} + \dots | s_{k} = s \Big]$$
  
=  $E \Big[ r_{k} + \beta V^{\pi}(s_{k+1}) | s_{k} = s \Big]$   
=  $\sum_{c \in C(s)} \pi(s, c) \Big[ r(s, c) + \beta \sum_{s'} p(s' | s, c) V^{\pi}(s') \Big]$  (3.1)

where  $\pi(s,c)$  is the probability of choosing  $c \in C(s)$  in state  $s \cdot V^{\pi}(s)$  is the value of state under policy  $\pi$  and  $V^{\pi}$  is called the state value function for  $\pi$ . The optimal state value function gives the maximum expected discounted utility at each state under an optimal policy:

$$V^{*}(s) = \max_{c \in C(s)} E[r_{k} + \beta V^{\pi}(s_{k+1})|s_{k} = s, a_{k} = a]$$
  
= 
$$\max_{c \in C(s)} \left[r(s,c) + \beta \sum_{s'} p(s'|s,c) V^{*}(s')\right]$$
(3.2)

Given an optimal state value function  $V^*$ , an optimal policy  $\pi^*$  can be constructed by selecting any choice c that yields the maximum in (3.2). That is setting  $\pi^*(s,c)=1.0$  and  $\pi^*(s,c')=0.0$  for any  $c' \neq c$ .

A similar value function for state-choice pair gives the expected utility of selecting any choice c in each state s under policy  $\pi$ :

$$Q^{\pi}(s,c) = E \Big[ r_{k} + \beta r_{k+1} + \beta^{2} r_{k+2} + \cdots | s_{k} = s, c_{k} = c \Big]$$
  
=  $E \Big[ r_{k} + \beta V^{\pi}(s_{k+1}) | s_{k} = s, c_{k} = c \Big]$   
=  $r(s,c) + \beta \sum_{s'} p(s'|s,c) \sum_{c'} \pi(s',c') Q^{\pi}(s',c')$  (3.3)

In addition, the optimal action value function is:

$$Q^{*}(s,c) = r(s,c) + \beta \sum_{s'} p(s'|s,c) \max_{c'} Q^{*}(s',c')$$
(3.4)

Boundary condition has to be specified to terminate the recursion in (3.1). Thus, (3.1) applies for all state, but

$$V(s) = 0, \quad s \in S_* \tag{3.5}$$

where  $S_*$  is a set of absorbing states in which all decisions have no effect. That is, when a traveller is in an absorbing state, all decisions immediately return to that state with a zero utility. For example, (T, Home) is a reasonable absorbing state in activity-travel scheduling.

To make sure travellers arrive home before or at T, V(s) is set to  $-\infty$  for any s such that  $t_s = T$  and  $a_s \neq \text{Home}$ . That is, the utility of making decisions that lead to these states is negative infinity. The utility of choosing to return home is zero, as specified in (3.5), but greater than any other choices. This is equivalent to imposing a constraint on decisions to prohibit any out-of-home activities at midnight. Actually, the latest hour of returning home can be any reasonable time of the day, such as 3:00am.

#### 3.2.2 Markov property

The *state* is composed of several components, such as the traveller's location (i.e., nodes in the transportation network) and the activity in which the traveller is engaged. The detailed specification of state is presented in Section 4.2.5.

The state of MDP model satisfies the Markov property. That is given the current state and choice,  $(s_k, c_k)$ , the subsequent state  $s_{k+1}$  is conditionally independent of

all the states and choices at time before k. In terms of conditional probability, Markov property is expressed as:

$$\Pr(s_k | s_{k-1}, c_{k-1}, \dots, s_1, c_1) = \Pr(s_k | s_{k-1}, c_{k-1})$$
(3.6)

Nevertheless, satisfying Markov property does not require the current decision to be irreverent to the past activity-travel choices. Two typical impacts of the past events on the successive decisions are discussed as follows.

For example, travellers are assumed to participate in any activity once and only once. Then the set of completed activities needs to be maintained and updated at each decision epoch. This dependence is handled by introducing an auxiliary set into the state. Further discussion is given in Section 4.2.5.

Another example is that, if a traveller drives to the office in the morning, the traveller must drive the car back home after work. The travel mode choice in the morning determines the choice after work. In this case, a binary variable is incorporated into the state to represent the availability of the car.

In summary, the impact of the past choices can be introduced into the state. The set of feasible choice at present depends on the current state. Thus, the state serves as an intermediate layer between the past choices and the successive choices. In other words, the past choices make an impact on the successive choices through the state.

## 3.2.3 Solution algorithms

The standard solution algorithms for solving the MDP model are value iteration and policy iteration (Puterman, 1994). The value iteration method is obtained by turning the Bellman optimality equation (3.1) into an update rule. Because of its simplicity, the value iteration method is adopted.

The value iteration method converges to an optimal solution for the discrete-time MDP with finite horizon. In practice, the algorithm stops once the expected utility changes by only a small amount in iteration. Figure 3.1 gives a complete value iteration algorithm with this termination criterion. The outputs of the algorithm are optimal decision  $\pi(s)$  and the expected utility  $V_k(s)$  for any feasible state s at iteration k.

for each state  $s \in S$  do set  $V_0(s) \leftarrow 0$ set  $k \leftarrow 0$ repeat set  $k \leftarrow k+1$ for each state  $s \in S$  do set  $V_k(s) \leftarrow \max_{c \in C(s)} \left\{ r(s,c) + \beta \sum_{s'} p(s'|s,c) V_{k-1}(s') \right\}$ until  $|V_k(s) - V_{k-1}(s)| < \epsilon$ for each state  $s \in S$  do set  $\pi(s) \leftarrow \underset{c \in C(s)}{\operatorname{arg}} \left\{ r(s,c) + \beta \sum_{s'} p(s'|s,c) V_{k-1}(s') \right\}$ return  $V_k$  and  $\pi$ 

Figure 3.1 Value iteration method for standard MDP model

## 3.3 An introductory example

A departure time choice problem is presented as follows to illustrate the essential ideas of MDP. Figure 3.2 shows a 2-node transportation network: W represents workplace and H represents home of a traveller. The traveller has to arrive home at or before 00:00am. The departure time choice is made to maximize the overall utility. If the traveller returns home, he will not go back to work.



Figure 3.2 A 2-node transportation network

Let  $\tau(t)$  denote the time needed to travel from W to H if the traveller departs at t o'clock. In this example, the travel time  $\tau(t)$  is fixed at I for any t = 1,...,12. In another word, travel time does not vary with departure time in this example. This assumption is made for simplicity and will be relaxed in Chapter 4 and 5. In addition, r(t,W) and r(t,H) denote the marginal utility obtained from one hour of work and in-home activity at t o'clock. Table 3.1 presents the marginal activity utility from 01:00pm to 00:00am.

Time	Time of the day	Marginal activity utility		
episode (t)		In-home $r(t, H)$	Work $r(t, W)$	
1	01:00pm	5 20		
2	02:00pm	5	20	
3	03:00pm	5	20	
4	04:00pm	5	20	
5	05:00pm	7.5	17.5	
6	06:00pm	10 15		
7	07:00pm	12.5 12.5		
8	08:00pm	15 10		
9	09:00pm	15	5 7.5	
10	10:00pm	15	5	
11	11:00pm	15	5	
12	12:00am	15	5	

Table 3.1 Marginal activity utility at each time episode

#### **3.3.1** Enumeration approach

Define U(t) as the overall utility that the traveller can obtain from 01:00pm to 00:00am if the work ends at t o'clock. Consequently, the overall utility is expressed as the sum of the utilities derived from work and in-home activity, and the travel disutility:

$$U(t) = \sum_{i=1}^{t} r(i, W) + \alpha \cdot \tau(t) + \sum_{i=t+\tau(t)+1}^{12} r(i, H)$$
(3.7)

where  $\alpha$  denotes the equivalent disutility of unit travel time. For demonstration, the value of  $\alpha$  is set to -5.0.

The optimal solution can be obtained by enumerating all the possible departure times. Figure 3.3 shows that the overall utility reaches the maximum value at 6 o'clock. When finer time scale is of interest and more choice dimensions are included, enumerating all the choice combinations is computationally impossible.



Figure 3.3 The overall utility for all possible departure times

## 3.3.2 Recursive method

Another approach is to formulate the departure time choice problem as a MDP model and solve it with a backward recursive method. At the beginning of each time episode, the traveller chooses the activity to be pursued in the following hour. The state has two components: the time episode and the traveller's location.

The backward recursive method starts at the last time episode. The only state that the traveller can have in the last time episode (12am) is at home. This is an absorbing state and thus, V(12, H) is set to zero. At 11:01pm, there are two possibilities: the traveller is at work or at home. If the traveller is at work, the only choice is to depart for home immediately so that he can arrive home at 12:00am. In this case, the utility obtained between 11:01pm and 00:00am is V(11, W) =  $-\alpha \cdot \tau(11) + V(12, H)$ . The other possibility is that the traveller already stays at home at 11:01pm. Then the traveller engages in in-home activity for the next hour. The utility derived from in-home activity between 11:01pm and 00:00am is

$$V(11,H) = r(11,H) + V(12,H)$$
(3.8)

The next solution step is to move back one time episode to 10:01pm. If the traveller has not returned home by 10:01pm, the traveller has two choices: continue at work or depart for home. The maximum utility that can be obtained in the next two hours is expressed as a maximization problem over the two choices:

$$V(10,W) = \max\begin{cases} r(10,W) + V(11,W), & \text{if continue to work} \\ \alpha \cdot \tau(10) + V(11,H), & \text{if depart for home} \end{cases} (3.9)$$

Similarly, the other possibility is that the traveller already stays at home at 10:01pm. The utility derived from in-home activity from 10:01pm to 12:00am is V(10, H) = r(10, H) + V(11, H).

Following the same procedure, the successive solution steps are performed by moving one time episode back in each step until the first time episode is reached. The entire solution process is described by recursive equations:

$$V(t,W) = \max\begin{cases} r(t,W) + V(t+1,W), & \text{if continue to work} \\ \alpha \cdot \tau(t) + V(t+\tau,H), & \text{if depart for home} \end{cases}$$
(3.10)

and

$$V(t,H) = r(t,H) + V(t+1,H)$$
(3.11)

#### 3.3.3 Random errors

In the real world, each traveller can have different perceptions of utility. To capture the perception error, a random term is incorporated into the utility, denoted by  $\varepsilon(t)$ .  $\varepsilon(t)$  is independently and identically distributed (i.i.d.). The expected maximum utility at t is expressed as:

$$V(t,W) = E\left[\max\begin{cases} r(t,W) + \varepsilon(t,W) + V(t+1,W)\\ \alpha \cdot \tau(t) + \varepsilon(t,H) + V(t+\tau,H) \end{cases}$$
(3.12)

If  $\varepsilon(t)$  is assumed an i.i.d extreme value random variable, (3.12) has the form of a logit model:

$$V(t,W) = \frac{1}{\theta} \log \left( e^{\theta \cdot (r(t,W) + V(t+1,W))} + e^{\theta \cdot (\alpha \cdot \tau(t) + V(t+\tau,H))} \right)$$
(3.13)

Suppose that there are 1,000 travellers. They choose departure time based on the activity utility and travel time given in Table 3.1. The perception parameter for extreme value random variable  $\varepsilon(t)$  is set to  $\theta = 0.1$ .

Figure 3.4 shows the expected maximum future utility at each time episode. From 1:00pm to 6:00pm, the expected maximum utility of *Work* is greater or equal to that of *Home*. From 7:00pm to midnight, travellers can obtain more utility by staying at *Home*.



Figure 3.4 Expected maximum future utility over time of the day

Figure 3.5 presents activity participation of the entire day. The population at *Home* and *Work* reflect the travellers' activity choices. The activity choices are determined by the utility profiles shown in Figure 3.4. The expected maximum future utility of staying at *Home* surpasses that of *Work* around 6:00pm. Travellers start to depart for *Home* at that time. Indicated by the light green bars in Figure 3.5, the departure flow from *Work* to *Home* reaches the maximum value at 7:00pm.



Figure 3.5 Activity participation of the entire day

## 3.4 Summary

The essentials of MDP modelling framework have been described in this chapter. The basic concepts of MDP, such as immediate utility, feasible state and decision, are illustrated and discussed in the introductory example.

The recursive method is more efficient than the enumeration approach, especially for larger transportation networks and multi-dimension choice (e.g. combined activity duration/timing choice and activity type/destination choice) problems. Let |D| denote the number of choices and |S| denote the number of states. The computational complexity of the recursive method is proportional to  $|D| \times |S|$ . The computational complexity of the enumeration approach is proportional to the number of all possible activity-travel schedules  $|S|^{|D|}$ .

# Chapter 4 Within-Day Dynamics in Individual's Activity-Travel Scheduling Behaviour

## 4.1 Introduction

The departure time choice model presented in Section 3.3 demonstrates an elementary application of Markov Decision Process (MDP). An MDP model of activity-travel scheduling, involving activity type, destination, timing and duration choices, is formally proposed in this chapter. The purposes of the proposed model are to capture within-day dynamics in activity-travel scheduling and thus, to replicate the traveller's daily activity-travel schedules.

The within-day dynamics in activity-travel scheduling include two types of behaviour. Firstly, the traveller takes into account the expected future utility at current decision and realizes the impact of the current decision on the future utility. This is the so-called forward-looking behaviour. Secondly, the activity-travel decisions have a strong dependency on contextual situations, such as time of day and location. This behaviour exhibits state dependence in the activity-travel scheduling.

In previous time-dependent activity-based network equilibrium models, travellers are assumed to care only about the current immediate utility and not take into account of the future utility (Huang and Lam, 2005; Lam and Yin, 2001). At each period, the traveller would take the choice with the highest immediate utility.

These assumptions are not consistent with the activity-travel behaviour in reality. Firstly, the traveller is able to adjust the daily activity-travel schedule to maximise the overall utility of the entire day. Second, the current decision partly determines the subsequent state of the traveller and thus, affects the utility that can be obtained in the near future. For example, if a traveller has done grocery shopping for food, the traveller will have enough inventory of food for consumption. Thus, the traveller needs not to go for shopping in the remainder of the day.

Following this line of reasoning, it is reasonable to assume that a traveller would take into account the future utility when making a decision at the current decision epoch. The traveller cannot foresee the exact future states and does not know the future utility with certainty. However, the traveller knows that in future decisions the activity-travel schedule can be adjusted to achieve the highest level of overall utility. Hence, although the traveller does not know the exact value of that future utility, he can make decisions based on the expectation of a maximum future utility. This consideration reflects the traveller's forward-looking behaviour.

The remainder of this chapter is organized as follows. Section 4.2 presents the activity-travel scheduling behaviour in the form of an MDP model. Section 4.3 describes the solution method for the MDP model. Section 4.4 illustrates the proposed model with numerical examples. The final section summarizes the key points of this chapter.

## 4.2 An individual's activity-travel scheduling model

The activity-based models commonly have hierarchical structures. These complicated structures can be divided into two levels. The upper level is concerned with activity participation over a planning horizon and generates alternative activity programs. Each activity program is composed of a set of activities to be completed during the planning horizon. For example, {*Home (before work), Work, Shopping, Home (after work)*} is an activity program for a typical weekday. The common activity programs can be extracted from activity-travel diary data. Activity programs used in the following discussion and numeric examples are hypothesized and designed for illustration purpose only.

The lower level is the implementation of the activity programs in a transportation network, involving multi-dimension choices, namely activity duration, destination and route choice. For each activity program, the multi-dimension choices in the lower level constitute a complete activity-travel schedule. For example, a daily activity-travel schedule can be, depart home at 8:00am, arrive at the office at 9:00am, work in the office until 6pm, have dinner outside and head for home at 9:00pm.

This chapter focuses on the within-day dynamics in activity-travel scheduling behaviour. The set of activity programs is assumed predefined. Other long-term choices, such as workplace and residential location, are treated as fixed and known.

## 4.2.1 Assumptions

Three assumptions are adopted in this chapter to regulate the individual's daily activity-travel schedules.

- A1. If a traveller has pursued an activity, the traveller will not re-engage in the activity within the planning horizon (Kasturirangan et al., 2002). Thus, each activity occupies a continuous period, represented by consecutive time episodes. The in-home activity is an exception, because a traveller may engage in the activity multiple times in a day. This exception can be handled by including multiple in-home activities in the activity set *A*. Specifically, two types of in-home activity are defined: the one occurring in the morning period Home-AM and the other in the evening period Home-PM.
- A2. If a traveller chooses to return home, the traveller must have completed all the out-of-home activities and will stay at home in the remaining time of the day. In other words, the activity-travel schedule of the traveller is a tour originating from home and terminating at home (Wen and Koppelman, 2000).

A3. A traveller terminates the current activity under two conditions: (i) voluntarily switch to another activity for higher overall utility, (ii) or exogenously be disrupted from the current activity because of some restrictions, such as supermarket opening and closing time. Thus, all out-of-home activities can only be pursued within a time interval, and let  $[\underline{t}_a, \overline{t}_a]$  denote the time interval in which activity  $a \in A$  can be pursued.

#### 4.2.2 Model formulation

Suppose that for a homogenous group of travellers there is a predefined set of activity programs N. Any activity program  $n \in N$  consists of a set of activities  $A^n$ . For simplicity, the superscript n is dropped. This does not cause any ambiguity, because all the formulas presented in this section apply to any activity program.

Time is discretised and a 24-hour day is evenly divided into T time episodes, denoted by  $\{1,...,T\}$ . The state s includes three variables that describe the traveller's contextual situations, including time of the day  $t_s$ , the current location of the traveller  $w_s$ , and the set of uncompleted activities  $A_s$ . Thus, the state s can be expressed as a 3-tuple  $(t_s, w_s, A_s)$ . An extensive discussion of the state transition is presented in Section 4.2.5.

At each decision epoch, a decision d is selected from a decision set D(s). There are two types of decisions, activity participation and travel. If d is a decision of activity participation, d is an ordered pair  $(a_d, h_d)$ , where  $a_d$  is the activity to be participated in and  $h_d$  is the chosen activity duration. Since the subsequent activity is a component of the decision, the order of activity participation is determined by the decision. If d is a decision of travelling, d is an ordered pair  $(z_d, m_d)$ , where  $z_d$  is the destination of the trip and  $m_d$  is the travel mode used to get to  $z_d$ . For the simplicity of the formulation, travelling is treated as a special activity. A travel decision  $(z_d, m_d)$  is also expressed in the form of  $(a_d, h_d)$ , where the chosen activity is  $a_d = \text{travel}$  and the activity duration is equal to the travel time  $h_d = \tau(t_s, w_s, z_d, m_d)$ .

Actually, the decision defined above persists for multiple time episodes and is different from the primitive choice with unit execution time in Section 3.2.1. Appendix I presents the formal derivation of decisions as a generalization of the primitive choices in standard MDP model. In the remainder of this thesis, the choices that affect the states and utilities for multiple time episodes are termed decisions.

Let  $D_{travel}(s)$  denote the set of travel decisions and  $D_{activity}(s)$  denote the set of activity decisions. The decision space is the union of the two sets  $D(s) = D_{activity}(s) \cup D_{travel}(s)$ . The definition of the decision set can be found in Section 4.2.4. The decisions over the entire day constitute a daily activity-travel schedule.

The immediate utility function is additively separable in the deterministic and random parts:

$$R(s,d) = r(s,d) + \varepsilon(d) \tag{4.1}$$

where  $\varepsilon(d)$  is a zero-mean random variable independently and identically distributed (i.i.d.) over travellers and states. The covariance structure of  $\varepsilon(d)$  is examined in Section 4.2.7.

The deterministic utility of a decision is the sum of the disutility of travelling or the activity utility. The disutility of travelling is determined by the departure time, the origin and destination of the trip. The amount of activity utility depends on the activity type, the activity timing and duration. Specifically, the total discounted utility of choosing d at state s is expressed as,

$$r(s,d) = \begin{cases} \sum_{k=1}^{h_d} \beta^{k-1} u(a_d, t_s + k) & \text{if } d \in D_{activity}(s) \\ \sum_{k=1}^{h_d} \beta^{k-1} \alpha_m & \text{if } d \in D_{travel}(s) \end{cases}$$
(4.2)

where  $\alpha_m$  is the equivalent disutility of unit travel time for travel mode m and u(a,t) is the marginal utility of activity a at time  $t \cdot h_d$  is the activity duration if  $d \in D_{activity}(s)$  and  $h_d = \tau(t_s, w_s, z_s, m_d)$  if  $d \in D_{travel}(s)$ .

At any decision epoch, a traveller makes a decision to maximize the weighted sum of the immediate utility and the expected future utility. The expected maximum utility is calculated by solving the recursive equation:

$$\overline{V}(s) = \mathbb{E}\left[\max_{d \in D(s)} \left\{ r(s,d) + \varepsilon(d) + \beta^{h_d} \cdot \overline{V}(s') \right\} \right]$$
(4.3)

where  $\beta \in [0,1]$  is the discount factor for future utility and is constant over time. Different values of  $\beta$  indicates a variety of behaviour patterns. If  $\beta = 0$ , the traveller is only concerned with immediate utility. If  $\beta = 1$ , the traveller places the same values on the immediate utility and the future utility of activities within the same day. As long as  $\beta > 0$ , the current decision depends on the future utility, This dependency reveals forward-looking behaviour.

In addition, the value of  $\beta$  can vary across individual travellers. In another word, each traveller may have distinct preference over future utility. The preference variation over population can be captured by a probability distribution function. If the choice probability of an individual traveller follows the logit model, the choice probability of the population is modelled by the mixed logit model (Hensher and Greene, 2003; McFadden and Train, 2000).

Assume that the random component of immediate utility  $\varepsilon$  are i.i.d. over states with cumulative distribution function  $G_{\varepsilon}(\cdot)$ . The expectation of the maximum utility conditional on state s is given by the integral:

$$\overline{V}(s) = \int \max_{d \in D(s)} \left\{ r(s,d) + \varepsilon(d) + \beta^{h_d} \cdot \overline{V}(s') \right\} dG_{\varepsilon}(\varepsilon)$$
(4.4)

Given the above assumptions of independence,  $\varepsilon$  is integrated out in above integral. Thus, the expected utility  $\overline{V}(s)$  is sufficient to represent the impact of  $\varepsilon$  on travellers' decisions. As the state is discrete, the expected maximum utility can be solved exactly, together with the decision-specific utility function expressed as:

$$\overline{v}(s,d) = r(s,d) + \beta^{h_d} \cdot \overline{V}(s')$$
(4.5)

For any  $s \in S$ , the conditional probability of choosing d is:

$$P(d \mid s) = \int I\left[\bar{v}(s,d) + \varepsilon(d) > \bar{v}(s,d') + \varepsilon(d'), \forall d' \in D(s)\right] \cdot dG_{\varepsilon}(\varepsilon) (4.6)$$

where  $G_{\varepsilon}(\cdot)$  is the cumulative distribution function of  $\varepsilon$ . I[·] is equal to one if the condition expression in the square bracket is true; otherwise, equal to zero.

## 4.2.3 Marginal utility

The traveller chooses the daily activity-travel schedule that provides the maximum overall utility. The traveller can schedule the start time and end time of an activity to gain as much utility as possible. The optimal starting time and the duration of activity depend on the temporal profile of activity utility. Hence, the activity-travel decision reflects the underlying temporal profile of activity utility. The bell-shaped marginal utility function presented by Ettema and Timmermans (2003) is employed in this thesis:

$$g_{a}(t) = U_{a}^{0} + \frac{\gamma_{a}\lambda_{a}U_{a}^{\max}}{\exp\left[\gamma_{a}\left(t - \xi_{a}\right)\right] \cdot \left\{1 + \exp\left[-\gamma_{a}\left(t - \xi_{a}\right)\right]\right\}^{\lambda_{a}+1}}$$
(4.7)

where t is time of the day,  $U_a^0$  is the baseline utility of activity a,  $U_a^{\text{max}}$  is the maximum marginal utility of activity a, and  $\gamma_a, \lambda_a, \xi_a, \eta_a$  are activity-specific parameters. The parameters can be calibrated from travel survey data. Figure 4.1 shows three examples of marginal activity utilities.



Figure 4.1 Marginal activity utilities of three activities over 24 hours

The marginal utility function is composed of two components: the baseline utility and the additional utility that defines the preference of activity timing. The baseline utility refers to the baseline preference for the activity. It is assumed to be a linear function of socio-economic and activity-specific variables and expressed as

$$U_{i,a}^0(x_i, x_a) = \mu_a x_i + \nu x_a$$

where  $x_i = (x_i^1, x_i^2 \cdots, x_i^J)$  is the vector of individual-specific variables defining the socio-economic status of traveller *i* and  $x_a = (x_a^1, x_a^2 \cdots, x_a^K)$  is the vector of activity-specific variables defining properties of activity a.  $\mu_a$  is the coefficient associated with  $x_i$  and v is the coefficient associated with  $x_a$ .

The utility of pursuing activity *a* in time interval  $[t+\delta h,t+\delta(h+1)]$  is calculated as:

$$u_a(t,h) = \int_{\delta h}^{\delta(h+1)} g_a(t+x,x) dx$$
(4.8)

To save computational time, the values of definite integral (4.8) can be precomputed and saved in a two-dimension array indexed by t and h.

#### 4.2.4 Decision set

If the traveller makes a travel decision, the decision has two components, the destination and the travel mode. Let B(z) denote the set of available activities at location  $z \in W$ . In particular, the residential location of the traveller is a location z such that  $\{\text{Home-PM}\} \subseteq B(z)$ . If  $A_s \cap B(z)$ , is an empty set, that is none of the uncompleted activities are available at z, location z cannot be chosen as the next activity destination. With these definitions, the set of travel decisions is expressed as:

$$D_{travel}(s) = \{(z,m) \mid z \in W \setminus \{w_s\}, A_s \cap B(z) \neq \emptyset, m \in M(w_s, z)\} \quad (4.9)$$

where  $M(w_s, z)$  is the available travel modes to go from  $w_s$  to z.

The decision can be augmented with additional components, such as choice of route  $y_r$  or choice of bus line  $y_b$ . Decision components  $y_r$  and  $y_b$  will be passed to the traffic assignment model and aggregated into traffic flows in road network or passenger flows in transit network. Particularly, the algorithm illustrated in Figure 4.3 is adopted to compute traffic flows and passenger flows based on  $y_r$  and  $y_b$ .

If the traveller makes an activity decision, the decision also has two components, the activity type and activity duration. According to the second assumption in Section 4.2.1, the traveller returns home if and only if all the out-of-home activities have been completed. Formally, let *s* denote the current state and  $A_s$  denote the set of uncompleted activities in state *s*. If there exists any uncompleted out-of-home activity, that is  $\{\text{Home}-\text{PM}\} \subsetneq A_s$ , Home-PM cannot be chosen and thus is excluded from the set. If Home-PM is the only one activity left in  $A_s$ , that is  $\{\text{Home}-\text{PM}\} = A_s$ , all the out-of-home activities have been completed and thus, Home-PM can be chosen. The set of feasible activities at state *s* is expressed as follows:

$$\overline{A}_{s} = \begin{cases} \left\{ a \mid a \in A_{s} \cap B(w_{s}), a \neq \text{Home-PM} \right\} & \{\text{Home-PM}\} \subsetneq A_{s} \\ \left\{ a \mid a \in A_{s} \cap B(w_{s}) \right\} & \{\text{Home-PM}\} = A_{s} \end{cases}$$
(4.10)

At the beginning of the day,  $A_s$  initially includes all the activities in the traveller's activity program. After that, at each decision epoch  $A_s$  is updated to exclude the completed activity. The formal update rule is presented in Section 4.2.5.

The third assumption in Section 4.2.1 requires that the starting time and the ending time of any activity a are within a given time interval  $[\underline{t}_a, \overline{t}_a]$ . Then the set of feasible activity decisions is defined as:

$$D_{activity}(s) = \left\{ (a,h) | [t_a, t_a + h] \subset [\underline{t}_a, \overline{t}_a], a \in \overline{A}_s \right\}$$
(4.11)

#### 4.2.5 Deterministic state transition

The state transition is assumed deterministic in this section. Thus, the travel time is deterministic and the subsequent state s' is uniquely determined by the current state s and the decision d. Consequently, the state transition probability function

takes a reduced form: there exists  $s' \in S$  such that p(s'|s,d)=1 and p(s'|s,d)=0 for any  $s' \neq s, s' \in S$ . This assumption however will be relaxed in Section 4.2.6 to incorporate travel time uncertainty into the model.

The time needed to execute decision d is equal to the activity duration  $h_d$  if  $d \in D_{activity}(s)$  and equal to the travel time  $\tau(t_s, w_s, z_d, m_d)$  if  $d \in D_{travel}(s)$ . Thus, the time of the day in the subsequent state s' is expressed by:

$$t_{s'} = \begin{cases} t_s + h_d & d \in D_{activity}\left(s\right) \\ t_s + \tau\left(t_s, w_s, z_d, m_d\right) & d \in D_{travel}\left(s\right) \end{cases}$$
(4.12)

If a travel decision is made, the location of the traveller is changed to the destination specified in the decision. Otherwise, the location remains unchanged:

$$w_{s'} = \begin{cases} w_s & d \in D_{activity}(s) \\ w_d & d \in D_{travel}(s) \end{cases}$$
(4.13)

According to the first assumption in Section 4.2.1, each activity is pursued only once within the planning horizon. Making an activity decision indicates that the current activity is completed. The completed activity is then excluded from the set of uncompleted activities  $A_{y}$ :

$$A_{s'} = \begin{cases} A_s \setminus \{a_s\} & d \in D_{activity}(s) \\ A_s & d \in D_{travel}(s) \end{cases}$$
(4.14)

In summary, the subsequent state is defined as follows:

$$s' = \begin{cases} \left(t_s + h_d, w_s, A \setminus \{a_s\}\right) & d \in D_{activity}\left(s\right) \\ \left(t_s + \tau\left(t_s, w_s, z_d, m_d\right), z_d, A\right) & d \in D_{travel}\left(s\right) \end{cases}$$
(4.15)

#### 4.2.6 Probabilistic state transition

Travel time is considered to have a stochastic nature. When travellers make activity-travel decisions, they can infer the traffic condition based on experience or using portable devices, such as smart phone and route guidance system. The travel time information obtained from these sources is not perfect. An unexpected change in travel time affects the activity-travel schedule. For example, if the travel time to a non-compulsory activity destination suddenly increases, the individual may cancel this activity and move on to another.

When travel time is uncertain, the arrival time is also uncertain and consequently, the subsequent state is a random variable. To capture the travel time uncertainty, the subsequent state is specified by a transition probability p(s'|s,d) rather than the deterministic transition presented in Section 4.2.5.

A discrete probability distribution is used to describe the stochastic travel time. For any  $j \in \mathbb{N}$ , let  $p_j(t_s, w_s, z_d, m_d)$  denote the probability of travel time j being realized. The state transition probability is equal to the probability that travelling from  $w_s$  to  $z_d$  by travel mode  $m_d$ :

$$p(s'|s,d) = p_j(t_s, w_s, z_d, m_d) \quad \text{if } s' \in \{x | t_x - t_s = j, j \in \mathbb{N}\} \quad (4.16)$$

With consideration of travel time uncertainty, the expected maximum utility is expressed by:

$$\overline{V}(s) = E\left[\max_{d}\left\{R(s,d) + \sum_{s'} p(s' \mid s,d) \cdot \beta^{h(s,s')} \cdot \overline{V}(s')\right\}\right]$$
(4.17)

where  $h(s, s') = t_{s'} - t_s$ .

## 4.2.7 Random errors

The random term in the immediate utility  $\varepsilon(d)$  is a vector of random variables with zero means. The dimension of  $\varepsilon(d)$  is determined by the number of alternatives in D(s). Let R(s,d) be the immediate utility of choosing d at state  $s \cdot \varepsilon$  enters the immediate utility in a separate and additive way:

$$R(s,d) = r(s,d) + \varepsilon(d)$$
(4.18)

where r(s,d) is the deterministic component and  $\varepsilon(d)$  is a zero-mean random variable.

In logit model,  $\varepsilon(d)$  is an i.i.d. Type I extreme-value random variable. Then the multi-dimension integral in (4.4) has a closed form expression:

$$\overline{V}(s) = \frac{1}{\theta_1} \log \left( \sum_{d \in D(s)} \exp(\theta_1 \cdot \overline{v}(s, d)) \right)$$
(4.19)

where  $\theta_1$  is a parameter related to the traveller's perception error and  $\overline{v}(s,d)$  is the decision-specific utility in (4.5). The probability of choosing *d* conditional on state *s* is expressed by:

$$P(d \mid s) = \frac{\exp(\theta_1 \cdot \bar{v}(s, d))}{\sum_{d' \in D(s)} \exp(\theta_1 \cdot \bar{v}(s, d'))}$$
(4.20)

It is well known that logit model suffers from the property of Independence of Irrelevant Alternatives (IIA). To capture the correlation between alternatives, a probit model is introduced to relax the limits on the covariance structure. The probit model avoids the IIA property by explicitly modelling the error covariance and thus allows flexible substitution among alternatives.

Using a probit model imposes a heavy computational burden. If there are J activity types and the maximum possible activity duration is T, an unrestricted covariance matrix of size  $TJ \times TJ$  is hard to estimate due to the identifiability

issue and computational burden. Thus, a structured covariance matrix is more tractable and interpretable.

The random term  $\varepsilon(d)$  is decomposed into two components,  $\varepsilon^1(d)$  and  $\varepsilon^0(d)$ .  $\varepsilon^1(d)$  is common for all decisions with the same activity type and  $\varepsilon^0(d)$  differs over activity types and activity durations.  $\varepsilon^1(d)$  and  $\varepsilon^0(d)$  are assumed independent from each other. The covariance of two decisions is then expressed as:

$$\operatorname{cov}(\varepsilon_{k}(d),\varepsilon_{k}(d')) = \operatorname{cov}(\varepsilon^{1}(d),\varepsilon^{1}(d')) + \operatorname{cov}(\varepsilon^{0}(d),\varepsilon^{0}(d')) (4.21)$$

where d = (a, h) and d' = (a', h') are decisions in D(s).

The first term in (4.21) is the covariance between different activity types. This term is defined as:

$$\operatorname{cov}(\varepsilon^{1}(d),\varepsilon^{1}(d')) = \sigma_{aa'}^{1}$$
(4.22)

where  $\sigma^1_{aa'}$  is constant and defined for every two activities.

It is reasonable to assume that the traveller's perception error is larger when the activity duration is longer. Thus, the second term in (4.21) increases with activity duration. The variance of  $\varepsilon^0(d)$  and  $\varepsilon^0(d')$  is then expressed as:

$$\operatorname{cov}\left(\varepsilon^{0}\left(d\right),\varepsilon^{0}\left(d'\right)\right) = \begin{cases} \min\left\{h,h'\right\}\cdot\sigma_{a}^{0} & a=a'\\ 0 & a\neq a' \end{cases}$$
(4.23)

where  $\sigma_a^0$  is a activity-specific constant.

Since  $\varepsilon$  follows a multivariate normal distribution, the evaluation of the expected maximum utility involves a multi-dimension integral, which the conventional numerical method cannot handle. A simulation method is needed to overcome this problem. The most widely used method is the GHK simulator (Train, 2003). The

maximum likelihood method for calibrating the parameters of the individual MDP model is briefly discussed in Section 6.2.

## 4.2.8 Relationship with activity-based supernetwork equilibrium models

As reviewed in Section 2.2.2, the supernetwork representation is used to model dynamic activity-travel choices in congested networks. The nodes in a supernetwork denote the possible time and location choices. The activity link indicates the process of activity participation for one time episode. The road link represents travelling between physical nodes.

Actually, the activity-based supernetwork equilibrium models are special cases of the proposed MDP model. Each node in a supernetwork can be represented by a state in the MDP model. Each link in the supernetwork is an activity or travel decision that connects one state to another. Each route in the supernetwork is an ordered set of states and decisions, which constitutes an activity-travel schedule. Each route begins with a state in time 0 and ends with a state in time T.

In the MDP model, the feasible schedules are defined by local rules for each state and decision. The number of states and decisions can be large, but the number of possible routes in a supernetwork is even larger. Directly imposing rules on routes makes the supernetwork models computationally intractable. The advantage of the MDP model is that it is much easier to define local rules for feasible states and decisions than to define rules for feasible routes in a supernetwork.

As discussed above, any supernetwork model can be represented by an equivalent MDP model. The supernetwork models are actually network equilibrium models with expanded networks. The MDP model can be integrated with traffic network equilibrium as well. The solution algorithms for the MDP model capturing activity-travel scheduling behaviour in a congestion network are developed in the next section.

## 4.3 Solution algorithms

Solution algorithms are developed to solve the MDP model and the network equilibrium problem simultaneously. The solution algorithms have a nested structure as shown in Figure 4.2.

The inner iteration is to find the optimal solution of the MDP model of the individual's activity-travel scheduling. Value iteration method presented in Section 3.2.3 is employed to solve the recursive equation (4.3). The solution includes the optimal values of  $\overline{V}(s)$  and the associated choice probability P(d | s).



Figure 4.2 Solution algorithms for the MDP model and the network equilibrium problem

The outer iteration is to find the equilibrium traffic flows by using the Method of Successive Averages (MSA). MSA is easy to be adapted for network equilibrium models and does not require computation of derivatives. The derivative-free feature of MSA is essential since the solution procedure of the MDP model involves an inner iteration and no closed-form solution exists.

The flow variables found in network equilibrium models, such as route flows, can be derived from the optimal values  $\overline{V}(s)$  and the optimal policy P(d | s). The steps of the derivation are presented as follows.

The expected maximum utility that can be obtained over an entire day by choosing activity program n is  $\overline{V}(s_0^n)$ , where  $s_0^n = (t_0, w_0, A_0)$  is the initial state.  $t_0$  is set to 1, the first time episode.  $w_0$  is set to the residential location.  $A_0$  is initialized to  $A^n$ , the set of all activities in activity program n.

If there are M travellers in the transportation system, the number of travellers choosing each activity program is assumed to follow a logit model:

$$M^{n} = M \cdot \frac{\exp\left(\theta_{2} \cdot \overline{V}\left(s_{0}^{n}\right)\right)}{\sum_{m=1}^{N} \exp\left(\theta_{2} \cdot \overline{V}\left(s_{0}^{m}\right)\right)}$$
(4.24)

where  $\theta_2$  is the perception error of choosing activity program.

Figure 4.3 show a breadth-first search algorithm for computing the number of travellers in state s, denoted by f(s). The algorithm starts from the initial state  $s_0^n$  and searches any states that can be reached from  $s_0^n$ . The variable f(s) is updated in each iteration. The algorithm terminates when all the states are visited.

for each state  $s \in S$  do set  $f(s) \leftarrow 0$ let Q be a first-in-first-out queue for each activity program  $n \in \{1, ..., N\}$  do set  $f(s_0^n) \leftarrow M^n$ push  $s_0^n$  into Qwhile  $Q \neq \emptyset$ pop the first element s from Q and remove s from Qfor each decision  $d \in D(s)$  do find the subsequent state s' using (4.15) if the time component t' of state s' > T then break out of the for loop else update  $f(s') \leftarrow f(s') + f(s) \cdot P(d \mid s)$ push s' into Q

Figure 4.3 Algorithm for calculating variables f(s)

If additional components, such as choice of route  $y_r$  or choice of bus line  $y_b$ , are incorporated into the decision, the aggregate traffic flows can be computed as follows. The traffic flow on a particular route  $y_r$  at time episode t is calculated by summing all the  $f(s) \cdot P(d | s)$  with route  $y_r$  as a component of decision d and  $t_s = t$ . Then the link flow on each road can be readily derived from the route flow. The number of passenger boarding on bus line  $y_b$  at stop w is calculated in a similar manner. The pseudo code of the algorithm is presented in Figure 4.4.

initialize route flows  $f(t, y_r) \leftarrow 0$  for any route  $y_r$ initialize variables  $f(t, y_b, w) \leftarrow 0$  for any bus line  $y_b$  and bus stop wfor each state  $s \in S$  do for each decision  $d \in D(s)$  do if mode choice  $m_d$  is private car update  $f(t_s, y_{d,r}) \leftarrow f(t_s, y_{d,r}) + f(s) \cdot P(d | s)$ if mode choice  $m_d$  is public transportation update  $f(t_s, y_{d,b}, w_s) \leftarrow f(t_s, y_{d,b}, w_s) + f(s) \cdot P(d | s)$ 

Figure 4.4 Algorithm for calculating flow variables

## 4.4 Numerical examples

In this section, the proposed MDP model and solution algorithm are applied to several numerical examples. The purpose of the examples is to analyse the individual's activity-travel scheduling behaviour of a homogenous group with 10,000 travellers in a period of 24 hours. Sensitivity analysis is conducted to examine how the computational results are affected by some key parameters of the model.

Figure 4.5 shows a small road network. There are three links connecting the locations of the three activities. The free travel time between any activity locations is labelled on the corresponding link. The dynamic link travel time is expressed as a simple Bureau of Public Road (BPR) function:

$$\tau_{l}(f_{l}(t)) = t_{l}^{0} \times \left(1 + 0.15 \left(\frac{f_{l}(t)}{5000}\right)^{4}\right)$$
(4.25)

where  $f_l(t)$  is the flow on link l at time t.

The equivalent disutility of travelling for one hour is  $\alpha = 60$ . The discount ratio of the future utility is set to  $\beta = 0.99$ . The random error in the immediate utility follows a i.i.d. extreme-value distribution with parameter  $\theta_1 = 0.2$ . The entire day (24 hours) is divided into 5-minute periods with 288 periods in total.



#### Figure 4.5 A 3-node network

There are three types of activity, namely staying at home, working and shopping. For simplicity, the choice of activity program is not explicitly considered. However, all the activity decisions effectively constitute the actual daily activity program. All the travellers are assumed to stay in home at 00:00 and have jobs with flexible working hours. Thus, travellers can fully control the duration of every activity and adjust the timing and duration to maximize the overall utility in the entire day. Table 4.1 presents the temporal constraints (e.g. opening hours) and parameters of marginal utility function for each activity.

Activities	Temporal constraints	Parameters of marginal utility function			
		Um	γ	λ	ξ (min)
Home	00:00 - 23:59	1000	0.006	1.0	0
Work	06:00 - 23:59	800	0.010	1.0	720
Shopping	10:00 - 23:59	180	0.032	1.0	1110

Table 4.1 Temporal constraints and parameters of marginal utility function

All the utility functions have single-peaked profiles as shown in Figure 4.6. The marginal utility of staying in home reaches the maximum value at midnight. The utility of working surpasses all the other activities at near 8:00 and remains so until the evening. At shopping time, travellers can achieve a high level of utility by consumption. Therefore, shopping has the steepest profile with a peak at 18:30.

The individual MDP model is clearly defined with all the above specifications. The model is implemented in AMPL, an algebraic modelling language for linear and nonlinear optimization problems (Fourer et al., 1990). The AMPL source code with comments is included in Appendix II. The computational results are presented as follows.



Figure 4.6 Temporal profiles of marginal utility functions

Figure 4.7 presents the number of travellers engaging in each activity over time of the day. Travellers start to leave home as early as 6:00. Most travellers arrive at office at around 8:00. The average duration of work is about 9 hours (from 8:00 to 17:00). After work, most travellers go for shopping and a small amount of traveller return home directly.

The pattern of activity participation illustrated in Figure 4.7 matches the temporal profiles of marginal utility functions depicted in Figure 4.6. This activity-travel pattern is used as the base scenario in the following sensitivity analysis. The remainder of the section presents sensitivity analysis of the key parameters, involving the discount ratio for future utility, the value of travel time and increase in travel time. The results obtained from the sensitivity analysis are compared with the base scenario.


Figure 4.7 Activity participation over time of the day

# 4.4.1 Discount ratio for future utility

Discount ratio  $\beta$  reveals how the travellers value the future utility. The value of  $\beta$  is related to the time scale. In this example, the time scale is 5 minutes and  $\beta = 0.99$ . That is a unit of utility obtained in time t + 1 is equivalent to 0.99 unit of utility obtained at time t. Generally, the utility obtained at a later time t + n is equivalent to  $0.99^n$  unit of utility at time t. For example, a unit of utility obtained one hour later is viewed as much as  $0.99^{12} = 0.886$  unit of utility at present. If  $\beta$  is set to 0.97, the discount ratio for utility obtained one hour later is 0.694. With a 5-minute time scale, even small variation in  $\beta$  can lead to a significant change in travellers' preference of future utility.

Figure 4.8 shows the variation in the timing of activity. The left part illustrates the activity participation over time of the day for  $\beta = 0.99$ , and the right part that for  $\beta = 0.97$ . Each activity is postponed as the value of  $\beta$  is decreased from 0.99 to 0.97. That is because if the discount ratio  $\beta \in [0,1]$  travellers always prefer to take a decision with greater utility as early as possible. Suppose that there are two tasks A and B that can be completed in a unit of time. The utilities that can be obtained

from A and B are 5 and 10, respectively. If A is completed before B, the total discounted utility is  $u_{AB} = 5 + \beta \cdot 10$ . If B is completed before A, the total discounted utility is  $u_{BA} = 10 + \beta \cdot 5$ . Since  $u_{BA} - u_{AB} = 5 - \beta \cdot 5 > 0$  for any  $\beta \in [0,1]$ , the individual prefers to complete B before A.

Switching from one activity to another involves travelling which causes disutility. Thus, according to the above argument, travellers try to postpone activity switching as much as possible. If the value of  $\beta$  decreases, travellers value future utility even less, and thus the timing of each activity is shifted to a later time.



Figure 4.8 Activity participation over time of a typical traveller ( $\beta = 0.99$  and  $\beta = 0.97$ )

### 4.4.2 Value of travel time

The value of travel time  $\alpha$  is the equivalent disutility of travelling for a unit of time. As the value of  $\alpha$  increases, the travellers tend to travel less. Travellers may even cancel their own activities to avoid the disutility of travelling. The value of

travel time thus affects the activity-travel pattern. Figure 4.9 shows the activity participation for different values of  $\alpha$ . The left part illustrates activity participation for  $\alpha = 60$  and the right part illustrates that for  $\alpha = 120$ . Most travellers go shopping for about 2 hours after work when they have a lower value of time. If the value of travel time is doubled, 55% of the travellers return home directly from the workplace and 45% half of the traveller go shopping for about 1 hour.



Figure 4.9 Activity participation of a typical traveller ( $\alpha = 60$  and  $\alpha = 120$ )

#### 4.4.3 Effect of increasing travel time

Travel time and the value of travel time jointly determine the travellers' perception of travel cost. However, road congestion does not only induce a longer travel time and a larger travel cost, but also reduces the time available to activity participation. This additional cost should be considered when evaluating transport policy. The individual's activity-travel scheduling model proposed in this chapter is capable of capturing the complicated interdependency between travel time and activity participation.

The link travel time depends on the traffic flows on the road. It is also affected by other external factors, such as weather condition and road construction. Suppose that such external factors cause the free travel time doubled. Table 4.2 shows that the time spent on shopping decreases from 2 hours to 0 hour and the time spent on in-home activity increases by 1.4 hours. The time that is originally intended for shopping is reallocated to in-home activity and travel.

Table 4.2 Allocation of time to activities and travel under different traffic conditions

Average duration (hours)	Home	Work	Shopping	Travel	Total
Normal traffic condition	11.6	9.3	2.0	1.1	24.0
Traffic congestion condition	13.0	9.1	0.0	1.9	24.0

Figure 4.10 shows the variation in activity participation due to the increasing travel time. The left part illustrates the activity participation for normal traffic condition. The right part illustrates that for traffic congestion condition. Compared with the normal condition, the travellers arrive at office much later due to the traffic congestion. To compensate for the late arrival at office, the travellers tend to leave office later in the evening. The working durations in both conditions are maintained at about 9 hours. The traveller gives up shopping and heads for home directly because of the postponed working hours and the longer travel time.



Figure 4.10 Activity participation of a typical traveller under different traffic conditions

# 4.5 Summary

In this chapter, a dynamic analytical model has been proposed to capture the state dependence and forward-looking behaviour in individual's daily activity-travel scheduling. The relationship between the current activity choice and the remaining uncompleted activities is explicitly considered. The model also captures travellers' choice of activity timing and duration in a congested network. Network equilibrium is incorporated into the MDP model of activity-travel scheduling. Solution algorithms are developed to solve the MDP model and the network equilibrium problem simultaneously.

The proposed model has four features. Firstly, time is treated as a scarce resource. Travellers allocate time to each activity to maximize the utility of the entire day. Second, a temporal utility function is employed in the model. The activity utility varies over time and affects the sequence of individual's activity participation. Third, exhaustive enumeration of all possible activity schedules is avoided by defining feasible decisions and states. The computational burden is thus greatly relieved. Finally, the individual's activity-travel scheduling behaviour is formulated as an MDP model. Hence, the results are consistent with the random utility maximization framework.

# Chapter 5 Intra-Household Interactions in Household's Activity-Travel Scheduling Behaviour

# 5.1 Introduction

Considerable evidence suggests that intra-household interactions play a crucial role in activity-travel scheduling. The individual's activity-travel scheduling model proposed in Chapter 4 ignores the interactions among household members and treats each individual separately. Further discussion is necessary to extend the proposed model to capture the intra-household interaction.

The activity-travel scheduling behaviour of a household with two members is investigated in this chapter. An analytical model is developed to explore how the interactions between household members influence the individual's activity-travel scheduling behaviour. Markov Decision Process (MDP) is employed to model household's scheduling behaviour and the intra-household interactions. MDP provides a modelling framework that allows the household's decisions to have complex interdependency over time.

The remainder of this chapter is organized as follows. Section 5.2 presents an extension of the individual's MDP model, which captures the intra-household interactions. Section 5.3 describes the solution algorithm for the extended MDP model. Section 5.4 illustrates the extended MDP model with numerical examples. The final section summarizes findings from this chapter.

# 5.2 An activity-travel scheduling model with intra-household interactions

An MDP model with intra-household interactions is presented in this section. The household is assumed to act as a single decision-making unit. The compulsory and non-compulsory activities of each household member are incorporated into the model by augmenting the state.

# 5.2.1 Assumptions

The following four assumptions are made to define the scope of the household activity-travel scheduling process.

- A1. The individuals in a household jointly make activity-travel decisions (Zhang et al., 2005). Each individual voluntarily takes actions to implement the decision.
- A2. Each individual has different preference over activity-travel schedules and the preference is represented by an individual utility function (Ettema et al., 2007).
- A3. Household members are honest in revealing their preferences. The individual preferences are therefore public information within the household.
- A4. With knowledge of individual preferences, the joint decision-making process seeks to maximize the welfare of the entire household (Bradley and Vovsha, 2004; Zhang et al., 2005).

Assumption A1 ensures that the joint decisions are effectively implemented by household members. With regard to Assumption A2, the individual utility functions can be used to derive the household utility function for activity-travel schedules. Assumptions A3 and A4 eliminate the possible strategic behaviour adopted by the individuals to gain advantages within the household. Without them, the household activity-travel scheduling process becomes a more general problem that game theory is needed to account for the individuals' strategic behaviour.

#### 5.2.2 Model formulation

The optimal activity-travel schedule of an individual traveller is well defined and solved in Chapter 4. However, a household often have multiple members. Each household member participates in activities in parallel. The combination of optimal decisions of all household members is not always optimal in terms of the welfare of the entire household. Typically, making an activity-travel decision for one household member constrains the decisions available for the other. For example, if a household member drives the only car of the household, the other member cannot choose private car as the travel mode.

For simplicity, a household with two members, indexed by  $i \in \{1, 2\}$ , is considered in this chapter. The model formulation can be easily generalized to any number of household members. Let  $A_i$  be the set of daily activities for individual i in the household. Each individual can undertake a subset of the activities in  $A_i$ . The activities in  $A_i$  are categorized into two types based on the flexibility of participation in Section 5.2.4.

The activity-travel scheduling behaviour of household member i is formulated as an individual MDP model, denoted by  $M_i$ . A subscript is used to distinguish the elements of an individual MDP model. For example,  $S_i$ ,  $D_i$ ,  $p_i$  and  $R_i$  are the state set, decision set, transition probability function and utility function of individual MDP model  $M_i$ . The two household members share the same discount ratio for future utility, denoted by  $\beta$ .

Formally, the activity-travel scheduling behaviour of the entire household is defined as a household MDP model. The state set of the household MDP is a proper subset of the cross product of the individual household members' state sets, i.e.,  $S = \{(s_1, s_2) | t_{s_1} = t_{s_2}, s_1 \in S_1, s_2 \in S_2\}$ . Notice that for any household state, the time episodes of the individual's states are synchronized.

The original individual's state defined in Chapter 4 is augmented with the on-going activity  $a_i$  and its remaining time  $e_i$ . The functionality of the additional components will be discussed later. The state of household member *i* is then a 5-tuple  $s_i = (t_{s_i}, w_{s_i}, A_{s_i}, e_{s_i})$ . Suppose that the state of a household member is (9AM, Office, Work, {Shopping, Home}, Work, 9 hours). This state represents that the household member works in the office at 9AM and will keep working for 9 hour. Two additional activities, Shopping and Home, need to be undertaken in the remainder of the day.

The constraints on the activity-travel decisions imply that the decision set of the household MDP model D, is a proper subset of the cross product of the individual household members' decision sets, i.e.,  $D \subseteq D_1 \times D_2$ . Each household member's decision set is defined as follows.

A household member's activity decision consists of the choice of activity type and duration:

$$D_{activity}(s_i) = \left\{ (a,h) \mid a \in A_{s_i}, [t_a, t_a + h] \subset \left[ \underline{t}_a, \overline{t}_a \right] \right\}$$
(5.1)

Suppose that the individual decision of a household member is (*Shopping, 1 hour*). This indicates that the household member will go shopping for 1 hour. Moreover, this activity has to be conducted in the opening hours.

A household member's travel decision consists of the choice of trip destination and travel mode:

$$D_{travel}\left(s_{i}\right) = \left\{\left(z,m\right) \mid A_{s_{i}} \cap B\left(z\right) \neq \emptyset, z \in W \setminus \left\{w_{s_{i}}\right\}, m \in M\left(w_{s_{i}}, z\right)\right\}$$
(5.2)

The union of the individual's travel decisions and activity decisions gives all the activity-travel decisions that the traveller can make:

$$D_{new}(s_i) = D_{activity}(s_i) \cup D_{travel}(s_i)$$
(5.3)

Each household member can participate in activities in parallel. The household is at a decision epoch whenever a household member has completed an activity. The other member, however, may have not completed his/her on-going activity. Formally, if the current state  $s = (s_1, s_2)$  is a decision epoch, each household member either takes a decision from  $D_{new}(s_i)$  or continues the on-going activity. A special decision set is defined to account for the on-going activity of individual household member  $D_{pre}(s_i) = \{(a_{s_i}, e_{s_i}) | e_{s_i} > 0\}$ . Then the set of feasible decisions for household member *i* is expressed as:

$$D(s_i) = \begin{cases} D_{new}(s_i) & s_i \in \mathcal{I}_i \\ D_{pre}(s_i) & s_i \notin \mathcal{I}_i \end{cases}$$
(5.4)

where  $\mathcal{I}_i = \{s_i \mid e_{s_i} = 0, s_i \in S_i\}$  is the set of decision epochs for individual *i*.

The set of feasible decisions for the entire household is the cross product of that of the two household members:

$$D(s) = D(s_1) \times D(s_2)$$
(5.5)

To allow the possibility of simply waiting for a household member to complete an on-going activity, the individual's decision set  $D(s_i)$  is augmented with a *wait* decision. The *wait* decision has a variable duration equal to the time until the next decision epoch. Travel decision is treated as a special activity *travel* with travel time as the activity duration.

Let  $Y(s,d) = \{(a_{d_i}, h_{d_i}) | d_i \in D(s_i), \forall i = 1, 2\}$  denote the set of the on-going activities and their remaining times until completion. The next decision epoch is the earliest time after which any on-going activity is completed:

$$\tau_d^s = \min_{(a,h)\in Y(s,d)} h \tag{5.6}$$

When a joint household decision  $d = (d_1, d_2)$  is made in state  $s = (s_1, s_2)$ , the subsequent state of household member *i* is updated as follows:

$$s_{i}' = \begin{cases} \left(t_{s_{i}} + \tau_{d}^{s}, w_{s_{i}}, A_{s_{i}} \setminus \{a_{s_{i}}\}, a_{d_{i}}, e_{s_{i}} - \tau_{d}^{s}\right) & d_{i} \in D_{activity}\left(s_{i}\right) \\ \left(t_{s_{i}} + \tau_{d}^{s}, z_{d_{i}}, A_{s_{i}}, travel, e_{s_{i}} - \tau_{d}^{s}\right) & d_{i} \in D_{travel}\left(s_{i}\right) \\ \left(t_{s_{i}} + \tau_{d}^{s}, w_{s_{i}}, A_{s_{i}}, a_{s_{i}}, e_{s_{i}} - \tau_{d}^{s}\right) & d_{i} \in D_{pre}\left(s_{i}\right) \end{cases}$$
(5.7)

The transition probability function of the household's state is defined as:

$$p(s'|s,d) = p_1(s_1'|s_1,d_1) \cdot p_2(s_2'|s_2,d_2)$$
(5.8)

The total discounted household utility of making joint household decision d at state s is expressed by:

$$r(s,d) = \sum_{k=1}^{\tau_d^*} \beta^{k-1} r(s,d,k)$$
 (5.9)

According to assumption A4 in Section 5.2.1, the household's objective is to make a joint household decision to maximize the expected overall utility:

$$\overline{V}(s) = \mathbb{E}\left[\max_{d \in D(s)} \left\{ r(s,d) + \varepsilon(d) + \beta^{\tau_d^s} \cdot \overline{V}(s') \right\} \right]$$
(5.10)

where  $\varepsilon(d)$  is the random error due to unobserved characteristics as explained previously in Section 4.2.7. The joint household decisions over the entire day constitute the daily activity-travel schedule of the household.

#### 5.2.3 Household utility function

A utility function is adopted to represent the household joint preference with consideration of intra-household interactions. The immediate utility that the household obtains at time k is decomposed as follows:

$$r(s,d,k) = \sigma_1 \cdot r_1(s_1,d_1,k) + \sigma_2 \cdot r_2(s_2,d_2,k) + r_J(s,d,k)$$
(5.11)

where  $\sigma_i$  is the weight parameter representing the relative influence of household member *i*.  $r_i$  is the individual utility that household member *i* can obtain when pursuing the activity independently:

$$r_{i}(s_{i},d_{i},k) = \begin{cases} \mu(a_{d_{i}},t_{s_{i}}+k) & \text{if } d_{i} \in D_{activity}(s_{i}) \\ \alpha(m_{d_{i}}) & \text{if } d_{i} \in D_{travel}(s_{i}) \end{cases}$$
(5.12)

 $r_J$  is the utility that the household can obtain for the joint activity participation:

$$r_{J}(s,d,k) = \rho \cdot r_{1}(s_{1},d_{1},k) \cdot r_{2}(s_{2},d_{2},k)$$
(5.13)

where  $\rho$  measures the level of interaction between household members' activities. The interaction coefficient  $\rho$  takes non-zero values if the two household members jointly pursue activity a at location w, i.e.,  $a_{s_1} = a_{s_2} = a$  and  $w_{s_1} = w_{s_2} = w$  for any state  $s \in S$ . Otherwise, (5.13) is equal to zero and the household utility (5.11) is reduced to the weighted sum of the individual utilities.

To model differences in intra-household interactions across activities, distinct interaction coefficient  $\rho_a$  can be specified for each activity a. Activities that require companionship and collaboration among household members have a positive intra-household interaction coefficient. Some routine activities that only need to be undertaken by any one of the individuals are specified with a negative  $\rho_a$ . That is the substitution between individuals exhibits negative intra-household

interaction. In summary, there exists positive interaction between household members if  $\rho > 0$ , negative interaction if  $\rho < 0$ , and no interaction if  $\rho = 0$ . Figure 5.1 depicts the household utility functions with distinct interaction coefficients.



Figure 5.1 The utility of joint activity participation for the entire household

The effect of the interaction coefficient can be further illustrated in an activity choice problem with two alternatives, A and B. Suppose that the household utility function takes a simple form:

$$r(a) = r_1(a) + r_2(a) + \rho \cdot r_1(a) \cdot r_2(a)$$
(5.14)

The individual utilities for the two activities are given in Table 5.1. Using (5.14) with  $\rho = 0$ , the household must be indifferent between A and B. If the household selects A, Individual 1 would receive higher utility from his preferred activity, while Individual 2 would receive lower utility from his less preferred activity. This result seems to be unfair to Individual 2. Thus, in order to avoid inequality, the

household should prefer B. When this is the case, this example indicates that  $\rho$  should take a positive value and thus, the household utility of choosing A is less than B, r(A) < r(B).

 Table 5.1 The utilities of independent activity participation

$r_i(a)$	Individual 1	Individual 2
Activity A	15	5
Activity B	10	10

Household members can choose the optimal individual decisions that are parts of the optimal household decision. However, there is no guarantee that the household members consider the same household decision, and thus that the actual household decision is in fact suboptimal for the entire household. Decision-making is even harder if there are multiple optimal household decisions. Coordination between household members is required to ensure that the optimal household decision is chosen.

### 5.2.4 Compulsory and non-compulsory activities

The daily activities can be categorized into two types based on the flexibility of participation, the compulsory and non-compulsory activities. All the compulsory activities are allocated to a specific household member and should be completed within the planning horizon. The non-compulsory activities are optional. Imposing these constraints on activity choice demonstrates the flexibility of the MDP framework.

To model the compulsory and non-compulsory activities, the original household state is augmented with an additional component  $G_s$ , the set of non-compulsory activities. Activities in  $G_s$  can be undertaken by any household member or be

skipped. The original set of daily activites  $A_{s_i}$  is redefined to include compulsory activities that must be completed by individual i.

The sets of non-compulsory activities in the subsequent state s' are updated as follows:

$$G_{s'} = \begin{cases} G_s & d_i \notin D_{activity}(s_i), \forall i \\ G_s \setminus \{a_{s_1}, a_{s_2}\} & \text{otherwise} \end{cases}$$
(5.15)

The other components of the subsequent state are updated according to state transition equation (5.7).

At each decision epoch, individual i can select a compulsory activity or a noncompulsory one. Thus, the individual's activity decision set is expressed as:

$$D_{activity}(s_i) = \left\{ (a,h) \mid a \in A_{s_i} \cup G_s, [t_a, t_a + h] \subset \left[ \underline{t}_a, \overline{t}_a \right] \right\}$$
(5.16)

To ensure that any individual completes the compulsory activities in  $A_{s_i}$ , for any absorbing state  $s \in S_*$ , the set of uncompleted compulsory activities should be empty,  $A_{s_1} = A_{s_2} = \emptyset, \forall (s_1, s_2) \in S_*$ .

# 5.3 Solution algorithms

Given the household MDP model and the optimal solutions of the individual's MDP models, one heuristic solution strategy is directly combining the optimal solutions of the individual's MDP models. Due to the intra-household interactions and constraints on household decisions, this strategy is suboptimal and even results in infeasible household decisions.

The following solution algorithm narrows down the household's decision space via dynamic merging the solutions of the individual MDP models (Singh and Cohn, 1998). The dynamic merging algorithm finds the optimal solution to the household

MDP model by directly performing value iteration on the household state and decision set. The pseudo code of the algorithm is presented in Figure 5.2. The equilibrium network flows can be computed by a nested method similar to the one presented in Figure 4.2. The only difference is that the household MDP model is solved by using the dynamic merging algorithm.

For a household MDP model,  $M = (M_1, M_2)$ , the individual MDPs  $M_1$  and  $M_2$ should be solved first using the algorithm presented in Section 3.2.3. Then the optimal values of  $M_1$  and  $M_2$ ,  $V(s_i)$ ,  $\forall s_i \in S_i$ , i = 1, 2 are used to construct the initial lower and upper bounds in the dynamic merging algorithm.

for each state $s \in S$ do					
set $(s_1, s_2) \leftarrow s$					
set $V_0(s) \leftarrow 0$					
set $V_0^L(s) \leftarrow \max\{V(s_1), V(s_2)\}$ and $V_0^U(s) \leftarrow V(s_1) + V(s_2)$					
set $k \leftarrow 0$					
repeat					
set $k \leftarrow k+1$					
for each state $s \in S$ do					
undate the lower and upper bounds.					
$V_k^L(s) \leftarrow \max_{d \in D_{k-1}(s)} \left\{ r(s,d) + \sum_{s' \in S} p(s' s,d) \cdot V_{k-1}^L(s' s,d) \right\}$					
$V_{k}^{U}(s) \leftarrow \max_{d \in D_{k-1}(s)} \left\{ r(s,d) + \sum_{s' \in S} p(s' s,d) \cdot V_{k-1}^{U}(s' s,d) \right\}$					
update the value of the household state:					
$V_k(s) \leftarrow \max_{d \in D_{k-1}(s)} \left\{ r(s,d) + \sum_{s' \in S} p(s' s,d) \cdot V_{k-1}(s' s,d) \right\}$					
update the set of competitive decisions:					
$D_k\left(s ight) \!\leftarrow\! \left\{d \in \! D_{k-1}\!\left(s ight)    Q_k^U\left(s,d ight) \!\geq\! V_t^L\left(s ight) ight\}$					
where $Q_{k}^{U}(s,d) = r(s,d) + \sum_{s' \in S} p(s' s,d) \cdot V_{k-1}^{U}(s' s,d)$					
<b>until</b> $ D_k(s)  = 1$ for all $s \in S$ or $ V_k(s) - V_{k-1}(s)  < \epsilon$					
for each state $s \in S$ do					
set $\pi(s) \leftarrow \underset{d \in D_k(s)}{\operatorname{argmax}} \sum_{s' \in S} F(s' s,d) [R(s,d) + V_k(s' s,d)]$					

return  $V_k$  and  $\pi$ 

Figure 5.2 Dynamic merging algorithm for household MDP model

The efficiency of dynamic merging is gained by constructing lower and upper bounds on the optimal values of the household states. The bounds are initially constructed based on the optimal solution of the individual MDP models and then incrementally updated using value iteration. If the upper bound of household decision d is less than the lower bound of another household decision, the decision d is strictly dominated and can be safely excluded from the household decision set. The algorithm terminates when there is only one household decision remaining in set  $D_k(s)$  for each household state s, or when the expected utility changes by a small amount in an iteration.

# **5.4 Numerical examples**

Figure 5.3 shows a 4-node road network on which activity-travel schedules are implemented. There are 10,000 behaviourally homogeneous households and each household is composed of two adults: Individual 1 and Individual 2. Node H represents the residential location. Node W1 and W2 represent the workplaces of Individual 1 and 2. For simplicity, travel time is assumed deterministic and the congestion effect is captured by a BPR function,

$$\tau_{l}(f_{l}(t)) = t_{l}^{0} \times \left(1 + 0.15 \left(\frac{f_{l}(t)}{5000}\right)^{4}\right)$$
(5.17)

where  $f_l(t)$  is the flow on link l at time t.

The equivalent disutility of travelling for one hour is  $\alpha = 60$ . The discount ratio of the future utility is set to  $\beta = 0.99$ . The entire day (24 hours) is divided into 5-minute periods with 288 periods in total.



Figure 5.3 A 4-node road network

The utility of pursuing an activity varies over the course of the day. The optimal starting time and the duration of activity depend on the temporal profile of activity utility. The bell-shaped marginal utility function proposed in (Ettema and Timmermans, 2003) is adopted in this example.

Three types of activity are considered in the example: *Home*, *Work*, and *Shopping*. The parameters of utility function for each activity are presented in Table 5.2. Figure 5.4 depicts the temporal profiles of the individual's marginal activity utility functions.

Activity	Parameters of utility function				
Activity	Um	γ	λ	ξ (min)	
Home	1000	0.006	1.0	0	
Work	800	0.010	1.0	720	
Shopping by Individual 1	180	0.032	1.0	1110	
Shopping by Individual 2	60	0.032	1.0	1110	

Table 5.2 Parameters of utility function for each activity

The two household members have distinct preferences for shopping activity. Individual 1 is more willing to go shopping than Individual 2. The distinct preferences of Individual 1 and 2 for shopping are represented by the bolded values of  $U_m$  in Table 5.2.



Figure 5.4 The individual's utility for independent activity participation with heterogeneous preferences

The welfare of the individuals in a household is treated equally important. The weight parameter,  $\sigma_i$ , representing the relative influence of household member *i* is thus assigned the same value:  $\sigma_1 = \sigma_2 = 1.0$ . The interaction coefficients of *Home* and *Work* are set to zero. The interaction coefficient of *Shopping* is denoted by  $\rho$ .

The household MDP model is defined by the above specifications. The solution algorithm in Section 5.3 is implemented in AMPL and used to solve the examples in this section. The source code is included in Appendix II. The model is examined for different types of intra-household interactions. The key findings are discussed as follows.

Initially, the interaction coefficient of *Shopping* is set to 0. Each individual makes activity and travel decisions independently. The activity-travel pattern of each individual reflects the underlying individual preference. Figure 5.5 illustrates the activity participation of Individual 1 and 2 over time of the day. Since Individual 1

gains a high level of utility from shopping, this individual goes shopping after work. Individual 2 prefers to return home directly after work.

The patterns of activity participation for Individual 1 and 2 depicted in Figure 5.5 are used as the base scenario for further analysis. The results of the positive and negative intra-household interactions are discussed and compared with the base scenario in the following sections.



Figure 5.5 Activity participation of a two-person household ( $\rho = 0$ )

# 5.4.1 Positive intra-household interaction for shopping

If a household considers shopping as a non-compulsory activity with positive intrahousehold interaction, the two household members will prefer to participate in the activity together to interact more with each other. Thus, the interaction coefficient should take a positive value,  $\rho = 0.2$  for example. The overall household's utility should be higher than the sum of individuals' utilities from independent activity participation. The extra utility received by the household is measured by (5.13).

Figure 5.6 shows the extra utility for different combinations of shopping timings and durations. For a given duration of shopping, an optimal timing gives the maximum utility. For any timing of shopping before 18:00, the utility increases rapidly with the duration of shopping. However, after 18:00 the gain of utility for spending an extra unit of time on shopping approaches zero. This tendency is illustrated by the contour lines parallel with y-axis between 18:00 and 19:00. This observation demonstrates that joint activity has an optimal timing and duration.



Figure 5.6 The utility of joint activity participation

Since joint participation provides a higher overall household utility than independent participation, the activity-travel pattern of Individual 2 is changed for the welfare of the household. Figure 5.7 shows that Individual 2 joins Individual 1 for shopping activity. On the other hand, the activity-travel pattern of Individual 1 does not show significant variation for the increase in  $\rho$ .



Figure 5.7 Activity participation of a two-person household ( $\rho = 0.2$ )

Table 5.3 presents the allocation of time to activities and travel for a large range of interaction coefficients. The duration of shopping for Individual 2 increases rapidly when  $\rho$  is increased from 0.0 to 0.2. However, this trend slows down when  $\rho$  approaches 0.5. Individual 2 spends less time on in-home activity to compensate the increased time in shopping. The working duration of Individual 2 is always maintained at 8 hours to 9 hours.

Values of p	Home	Work	Shopping	Travel	Total
0.0	13.8	9.4	0.0	0.8	24.0
0.2	11.9	9.2	1.8	1.1	24.0
0.5	11.7	8.9	2.3	1.1	24.0
1.0	11.6	8.4	2.9	1.1	24.0

Table 5.3 Allocation of time to activities and travel for different values of  $\rho$  (Individual 2)

#### 5.4.2 Negative intra-household interaction for shopping

If the household considers shopping as a non-compulsory activity with negative intra-household interaction, only one household member will take action to complete the shopping activity and the entire household benefits from that action. If a household member has done the shopping task, the benefit of another shopping trip is negligible and the cost of the trip is significant, particularly in a congested transportation network.

The action of any household member is thus substitutable within the household. The interaction coefficient takes a negative value in this case. Figure 5.8 depicts the utility of independent activity participation with homogenous individual preferences over shopping.



Figure 5.8 The utility of independent activity participation with homogenous preferences

The simulated time-space path in Figure 5.9 shows the possible activity-travel patterns. The two individuals share the same temporal profile of marginal utility for shopping. The household can assign the shopping task to either individual and get the same overall utility. The probability of going shopping for any individual is 50%.



Figure 5.9 Activity participation over time of the day ( $\rho = -0.2$ )

# 5.5 Summary

A household MDP model with consideration of the intra-household interactions has been proposed to describe household members' daily activity-travel scheduling behaviour. The proposed model allows decomposing the household's utility into two components: the utility of engaging in an activity independently and the utility derived from the joint activity participation with other household members. The model also enables complicated activity-travel decisions over time.

The results drawn from the numerical analysis are summarized as follows. Firstly, household members may have distinct preferences for the same activity. If the intra-household interaction for an activity is zero, each household member chooses activity-travel schedule based on the individual utility. Secondly, the strength of

the intra-household interaction varies across activity types. If the intra-household interaction is positive for an activity, household members participate in the activity jointly to obtain higher household utility. If the intra-household interaction is negative for an activity, this activity can be assigned to either household member. The household utility is maintained at the same level.

# **Chapter 6** Calibration Methods and Results

# 6.1 Introduction

This chapter aims to introduce statistical methods for calibrating the parameters of the MDP models proposed in Chapter 4 and Chapter 5. Time-series data are required for calibration of model parameters. The dataset should include travellers' activity choices and geographic locations over time. Due to the cost of collecting time-series activity-travel data, hypothetical numerical experiments are conducted to generate the necessary dataset.

This chapter is organized as follows. Section 6.2 presents the formulation of maximum likelihood method. Section 6.3 reviews two numerical methods for solving the calibration problem. Section 6.4 presents the data generated from numerical experiments and the calibration results.

# 6.2 Maximum likelihood method

Rust (1994) proposed a unified framework for calibration of MDP model. The contribution of individual i to the log-likelihood function is expressed as follows:

$$l_{i}(\eta) = \sum_{t=1}^{T_{i}} \log P(d_{it} \mid s_{it}, \eta) + \sum_{t=1}^{T_{i}-1} \log p(s_{i,t+1} \mid d_{it}, x_{it}, \eta) + \log \Pr(s_{i1} \mid \eta)$$
(6.1)

where  $\eta$  represents the vector of model parameters and  $p(s_{i,t+1} | d_{it}, x_{it}, \eta)$  is the state transition probability function conditional on  $d_{it}, x_{it}$  and  $\eta$ .

Assume that  $\varepsilon$  is normally distributed with  $N \times M$ -variate probability density function  $G_{\varepsilon}(\cdot)$ . The probability of observing choice  $d_i$  can be calculated by integrating over all possible values of  $\varepsilon$ ,

$$P(d_{it} | \eta) = \int \cdots \int G_{\varepsilon}(\varepsilon) d\varepsilon$$
(6.2)

If the choice is independent over individuals, the likelihood of all individuals' activity choices can be expressed as the product of each individual's activity choice probability:

$$L(d_{1},...,d_{I} | \eta) = \prod_{i=1}^{I} l_{i}(\eta)$$
(6.3)

As the choice probability involves multi-dimensional integral, (6.3) is evaluated using the GHK simulator. Consistent results can be obtained by the simulated likelihood method.

# **6.3** Numerical methods for parameter calibration

The MDP model is traditionally calibrated by using Nested Fixed-Point (NFXP) algorithm (Rust, 1988, 1987). However, this method is computational demanding and is thought to be impractical in many contexts. Formulating the calibration problem as Mathematical Programming with Equilibrium Constraints (MPEC) greatly reduce the computational burden (Su and Judd, 2008). This study thus adopts the MPEC approach for solving the maximum likelihood problem.

# 6.3.1 Nested fixed-point algorithm

The NFXP algorithm finds solutions in a nested manner. An inner fixed-point algorithm computes the unknown endogenous variables for each value of model parameter. In the activity-travel scheduling models, the endogenous variables are the travellers' decisions over time. An outer hill climbing algorithm searches for the model parameter that maximizes the likelihood function. The NFXP algorithm is an intuitional and natural method of implementing the maximum likelihood method. The drawback is the computational burden of solving the dynamic programming problem thousands of times in the inner loop. Even though the original author implemented the algorithm in GAUSS language and the program is in the public domain, the code has not been updated for years. The researcher has to implement the algorithm from scratch if it is adopted to solve the calibration problem.

# 6.3.2 Mathematical programming with equilibrium constraints

The idea behind the Mathematical Programming with Equilibrium Constraints (MPEC) approach is simple. It aims to search the model parameters and endogenous variables to maximize the likelihood function subject to the equilibrium constraint. The endogenous variables fulfil the equilibrium condition defined by the model parameters. The researcher can simply write down the likelihood function and the equilibrium constraints in algebraic modelling languages. Then the model parameters are calibrated with the state-of-the-art constrained nonlinear optimization solvers. The calibration of individual and household activity-travel scheduling models is implemented in AMPL and solved with KNITRO. The AMPL source code can be found in Appendix II.

The NFXP algorithm solves the dynamic programming problem with high accuracy for each guess of the model parameters. In contrast, most modern solvers of MPEC only need to solve the dynamic programming problem at the final iteration for calculating the results. The computational burden is greatly reduced by this strategy. Su and Judd (2008) showed that if MPEC and NFXP are used to solve the same calibration problem, the two methods yield the same calibration results.

# 6.4 Calibrating the individual's MDP model

The calibration of parameters in individual's activity-travel scheduling model is presented in this section. For a given set of parameters, Monte Carlo experiments were conducted to generate time-series data. Based on the generated data, the MPEC approach was then employed to calibrate the model parameters. The difference of the actual parameters and the calibrated parameters is used to evaluate the accuracy of the calibration method.

# 6.4.1 Marginal utility functions

The first marginal utility function is a bell-shaped function (Ettema and Timmermans, 2003):

$$g_{a}(t) = \frac{\gamma_{a}\lambda_{a}U_{a}^{\max}}{\exp\left[\gamma_{a}\left(t-\xi_{a}\right)\right]\cdot\left\{1+\exp\left[-\gamma_{a}\left(t-\xi_{a}\right)\right]\right\}^{\lambda_{a}+1}}$$
(6.4)

The second marginal utility function is based on a scaled probability density function of the scaled Cauchy distribution (Ettema et al., 2004):

$$g_{a}(t) = \frac{U_{a}^{\max}}{\pi c_{a} \left[ 1 + \left( \frac{t - b_{a}}{c_{a}} \right)^{2} \right]}$$
(6.5)

where  $U_a^{\text{max}}$  is the maximum marginal utility,  $b_a$  is the time at which the marginal utility reaches the maximum value and  $c_a$  determines the period in which a satisfactory marginal utility can be obtained.

Figure 6.1 depicts the temporal profiles of these two marginal utility functions. Both functions are unimodal (having a single local maximum) and ensure that the marginal activity utility increases in the warm-up period and decreases in the saturated period. Although the shapes of the two curves are similar, the scaled Cauchy distribution has a sharp peak and a long tail. Essentially, either marginal utility function can be adopted for empirical analysis. However, the scaled Cauchy distribution has fewer parameters and a simpler functional form. Thus, the scaled Cauchy distribution has fewer identifiability problems and it is employed in the following numerical experiment.



Figure 6.1 Temporal profiles of two types of marginal utility function

# 6.4.2 Activity-travel data generation

Figure 6.2 shows a transportation network with 3 nodes. Each node represents an activity destination. The free flow travel time of each link is given in the figure. The travel time in congestion is captured by a BPR function:

$$t_{l}(f_{l}(t)) = t_{l}^{0} \times \left(1 + 0.15 \left(\frac{f_{l}(t)}{5000}\right)^{4}\right)$$
(6.6)

where  $t_l^0$  is the free flow travel time and  $f_l(t)$  is the flow on link *l* at time *t*.



Figure 6.2 A 3-node transportation network

The time is evenly divided into 5-minute intervals. The value of time is set to 60. The discount ratio of the future utility is set to 0.95. These two parameters are treated as fixed and known to maintain the identifiability of the problem.

The marginal activity utility varies over time and is defined by marginal utility function (6.5). Table 6.1 presents the actual values of the parameters of the marginal utility function. These are the parameters to be calibrated. The calibration results will be compared with the actual values in the following.

No	Activity types	Parameters					
1	Home-AM/PM	$U_1^m$	3600	$b_1$	0	$c_1$	320
2	Work	$U_2^m$	2500	$b_2$	840	$c_2$	180
3	Shopping	$U_3^m$	2000	$b_3$	1140	<i>C</i> <sub>3</sub>	210

Table 6.1 Parameters of marginal utility function

Figure 6.3 depicts the shape of the marginal utility function. There are two in-home activities: *Home-AM* and *Home-PM*. The parameters of the marginal utility functions of the two activities are the same as shown in Table 6.1. From 00:00 to 08:00, *Home-AM* is the activity with the maximum marginal utility. From 08:00 to 17:00, *Work* dominates the other activities in terms of marginal utility. From 17:00 to 21:00, *Shopping* is the dominate activity. From 21:00 to midnight, *Home-PM* is the dominate activity.



Figure 6.3 Temporal profile of the marginal utility function

The travellers are assumed to choose the daily activity program, activity duration and departure time to maximize the overall utility of the entire day. Their utility maximization behaviours are described by the MDP model. Under these assumptions, time-series data for 288 time intervals (24 hours) and 10000 travellers were generated in the numerical experiment. The choice probability is assumed to follow equation (4.20) and the parameter  $\theta$  is fixed at 0.2.

The Monte Carlo method was conducted as follows: (1) fix the model parameters at actual values and solve the Bellman equation (4.4) to obtain the optimal value of  $\overline{V}(s), s \in S$ ; (2) use the actual values of the model parameters and  $\overline{V}(s)$  to compute the conditional choice probability (4.6); (3) generate choices and state transitions for 10000 travellers in 288 periods based on the choice probability and the travel time.

# 6.4.3 Calibration results

Before reporting the calibration results, the profile of log-likelihood function and the maximum values are illustrated and discussed. Given the activity-travel data, the log-likelihood function depends on a vector of parameters. Visualizing a multidimensional function is hard. This section thus seeks to illustrate the impact of one or two parameters of interest on the log-likelihood function.

The parameters of marginal utility function can be represented as a vector,  $\eta = (U^{\max}, b, c)$ , where  $U^{\max} = (U_1^{\max}, U_2^{\max}, U_3^{\max})$ ,  $b = (b_1, b_2, b_3)$ ,  $c = (c_1, c_2, c_3)$ . Let  $\eta$  be the overall maximum likelihood estimate of  $\eta$  and  $\eta(b_2)$  be the vector of parameters with all the parameters except  $b_2$  fixed at the maximum likelihood estimate of  $\eta$ . Then the log-likelihood function is defined by:

$$l_i(b_2) = l_i(\eta(b_2)) \tag{6.7}$$

Figure 6.4 illustrates the log likelihood  $l_i(b_2)$  as a function of  $b_2$  with other parameters fixed at the optimal values. The log-likelihood function  $l_i(b_2)$  is nonconvex and has a unique maximum value at  $b_2 = 841.4$ , which is very close the true value 840. If  $b_2$  is shifted a little from the optimal value, the value of the log likelihood changes dramatically. Vector *b* determines the time at which the marginal utility function reaches the maximum value and thus, has a strong influence on the activity timing choice.



Figure 6.4 The log-likelihood function l<sub>i</sub>(b<sub>2</sub>)

Let  $\eta(b_2, b_3)$  be the vector of parameters with all the parameters except  $b_2$  and  $b_3$  fixed at the maximum likelihood estimate of  $\eta$ . Then the log-likelihood function is defined by:

$$l_i(b_2, b_3) = l_i(\eta(b_2, b_3))$$
(6.8)

Figure 6.5 depicts the log likelihood as a function of  $b_2$  and  $b_3$ . The overall appearance of the log-likelihood function reveals a rather complicated relationship between the log likelihood and the model parameters  $b_2$  and  $b_3$ . Multiple local optimal solutions can be found in the figure.



Figure 6.5 Contour and 3-D plot of the log-likelihood function  $l_i(b_2, b_3)$ 

Table 6.2 presents the calibration results of the model parameters. In general, the relative errors of the calibrated values are within 10%. The calibrated maximum marginal utility  $U^{\text{max}}$  is smaller than the actual value. The calibrated location parameter *b* is very close to the actual values. *b* has a greater impact on the dominate period of each activity than  $U^{\text{max}}$  and thus to a great extent determines the activity choice probability and the log likelihood. This is the reason that the calibration of *b* is more accurate than that of  $U^{\text{max}}$ . Similarly, the same argument

applies to the calibration of c, which determines the width of the marginal utility curve.

 $U^{\max}$  $\hat{c}$  $\hat{b}$ Parameters W Η W S S Η W S Activity types Η Actual values 3600 2500 2000 0 840 1140 320 210 180 Calibrated values 3517 2481 2051 0 841 1140 305 168 219

Table 6.2 Calibration results of parameters in the marginal utility function

# 6.5 Calibrating the household's MDP model

This section presents the calibration of the intra-household interaction coefficient. The household's MDP models include two types of parameters, the intra-household interaction coefficients and the parameters of marginal utility function. Household members have distinct preference over the timing of activities and thus, the parameters of marginal utility function for each member are defined and calibrated.

# 6.5.1 Household activity-travel data generation

Figure 6.6 shows a 4-node road network on which activity-travel decisions are made. There are 10,000 behaviourally homogeneous households and each household is composed of two adults: Individual 1 and Individual 2. Node H represents the residential location. Node W1 and W2 are the workplaces of the household members respectively. For simplicity, travel time is assumed deterministic and the congestion effect is captured by a BPR function,

$$\tau_{l}(f_{l}(t)) = t_{l}^{0} \times \left(1 + 0.15 \left(\frac{f_{l}(t)}{5000}\right)^{4}\right)$$
(6.9)

where  $f_l(t)$  is the flow on link l at time t.
The equivalent disutility of travelling for one hour is  $\alpha = 60$ . The discount ratio of the future utility is set to  $\beta = 0.95$ . The entire day (24 hours) is divided into 5-minute periods and there are 288 periods in total.



Figure 6.6 A 4-node road network

The scaled Cauchy distribution proposed in (Ettema and Timmermans, 2003) is adopted as the marginal utility function in this example. Three types of activity are considered in the example: *Home, Work,* and *Shopping.* The parameters of utility function for each activity are presented in Table 6.3. The two household members have distinct preferences for work and shopping activity. Individual 1 is more willing to go shopping than Individual 2, but receives less utility from work (represented by the bold values in Table 6.3). Figure 6.7 depicts the temporal profiles of the individual's marginal activity utility functions.

	Activity	Parameters of utility function							
No		1	ndividual	l	Individual 2				
		$U^{m}$	b (min)	С	$U^{m}$	b	С		
1	Home-AM/PM	3600	0	320	3600	0	320		
2	Work	2500	840	180	3000	840	180		
3	Shopping	2000	1140	210	1500	1140	210		

Table 6.3 Parameters of marginal utility functions for the household



Figure 6.7 Temporal profile of the marginal utility function

The marginal utility for a household defined by equations (5.11) and (5.13) are replicated as follows.

$$\boldsymbol{r}_a = \boldsymbol{\sigma}_1 \cdot \boldsymbol{r}_{1,a} + \boldsymbol{\sigma}_2 \cdot \boldsymbol{r}_{2,a} + \boldsymbol{\rho}_a \cdot \boldsymbol{r}_{1,a} \cdot \boldsymbol{r}_{2,a}$$
(6.10)

where  $r_{i,a}$  is the individual utility that household member *i* can obtain when pursuing the activity *a* independently. The welfare of the individuals in a household is treated equally important. The weight parameter  $\sigma_i$ , representing the relative influence of household member *i*, is thus fixed at the same value,  $\sigma_1 = \sigma_2 = 1.0$ . The interaction coefficient Work  $\rho_2$  is set to zero. The interaction coefficient of *Home* and *Shopping* is set to  $\rho_1 = 0.3$  and  $\rho = 0.2$ .

#### 6.5.2 Calibration results

The main focus of this section is to show how to calibrate the intra-household interaction coefficient  $\rho$ . The log-likelihood function is a multidimensional function of a vector of parameters and hard to be visualized. Following the approach presented in Section 6.4.3, the impact of one or two parameters of interest, i.e., the interaction coefficients, on log-likelihood function is visualized and discussed.

The parameters of the household's MDP model can be represented as a vector,  $\eta = (U_1^{\max}, b_1, c_1, U_2^{\max}, b_2, c_2, \rho)$ , where  $U_i^{\max}$ ,  $b_i$  and  $c_i$  are the vectors of parameters defined for household member *i*'s marginal utility function, and  $\rho$  is the vector of intra-household interaction coefficients defined for *Home*, *Work*, and *Shopping*,  $\rho = (\rho_1, \rho_2, \rho_3)$ .

Denote by  $\eta$  the overall maximum likelihood estimate of  $\eta$  and let  $\eta(\rho_3)$  be the vector of parameters with all the parameters except  $\rho_3$  fixed at the maximum likelihood estimate of  $\eta$ . Then the log-likelihood function is defined by:

$$l_i(\rho_3) = l_i(\eta(\rho_3)) \tag{6.11}$$

Figure 6.8 illustrates the log likelihood as a function of  $\rho_3$  with other parameters fixed at the calibrated values. The only local maximum of log-likelihood function  $l_i(\rho_3)$  is also a global maximum.  $l_i(\rho_3)$  has the global maximum at  $\rho_3 = 0.188$ . The figure shows that log-likelihood function  $l_i(\rho_3)$  is concave in interval [0,1]. However, no formal proof is obtained to confirm this observation.



Figure 6.8 The log-likelihood function  $l_i(\rho_3)$ 

Similarly, given the activity-travel data, the log likelihood can be defined as a function of  $\rho_1$  and  $\rho_3$ , i.e.,  $l_i(\rho_1, \rho_3)$ . Figure 6.9 shows that the log-likelihood function  $l_i(\rho_1, \rho_3)$  is concave in the unit square  $[0,1]^2$  and has a global maximum at point  $(\rho_1, \rho_3) = (0.282, 0.188)$ .



Figure 6.9 Contour and 3-D plot of the log-likelihood function  $l_i(\rho_1, \rho_3)$ 

The true value of intra-household interaction coefficient  $\rho$  is (0.3,0.0,0.2) and the calibrated value of  $\rho$  is (0.282,0.011,0.188). The accurate calibration of  $\rho$ can be contributed to the concaveness of the log-likelihood function over the unit cube  $[0,1]^3$ . Table 6.4 presents the calibration results of the parameters in marginal utility functions.

The household utility function (6.10) is symmetric with respect to individual's utilities  $r_{1,3}$  and  $r_{2,3}$ . Therefore, the respective calibrated values of  $U_{1,3}^{\text{max}}$  and  $U_{2,3}^{\text{max}}$  have little effect on the household utility as long as their product is comparable to that of the true values. As shown in Table 6.4, the calibrated values of  $U_{1,3}^{\text{max}}$  and  $U_{2,3}^{\text{max}}$  are 1791 and 1757, and their product is 3146787. The true values of  $U_{1,3}^{\text{max}}$  and  $U_{2,3}^{\text{max}}$  are 2000 and 1500 and their product is 3000000. This

explains why the calibration results of these two parameters have less accuracy.

The relative errors of the other calibrated values are within 10%.

Household	Parameters	$U^{\max}$		$\hat{b}$			ĉ			
members	Activity types	Н	W	S	Н	W	S	Н	W	S
1	Actual values	3600	2500	2000	0	840	1140	320	180	210
1	Calibrated values	3654	2389	1791	1	828	1149	338	178	196
2	Actual values	3600	3000	1500	0	840	1140	320	180	210
2	Calibrated values	3667	3114	1757	0	849	1128	305	182	226

Table 6.4 Calibration results of parameters in the marginal utility function for each household member

## 6.6 Summary

This chapter presents the maximum likelihood method for calibrating the MDP model of individual and household activity-travel scheduling. The activity-travel data required for calibration were generated from numerical experiments. The calibration method was tested and evaluated with these hypothetical data. The calibration results were found satisfactory and the relative errors of most results are within 10%. Notably, numerical experiments reveal that the log-likelihood function is concave over the domain of intra-household interaction coefficient. This property enables efficient and accurate calibration of the coefficient.

# **Chapter 7** Conclusions and Discussion

### 7.1 Summary and conclusions

In this thesis, an activity-based network equilibrium modelling framework has been proposed to investigate within-day dynamics and intra-household interactions in activity-travel scheduling behaviour. The activity-travel scheduling behaviour is formulated as Markov Decision Process (MDP). The advantage of the MDP model is that constraints can be imposed on the activity-travel decisions without increasing the computational burden. The MDP model is extended to take into account intra-household interactions for compulsory and non-compulsory activities. Complicated interdependency between household member's activitytravel decisions over time is enabled because of the flexibility of the MDP model.

The individual MDP model has been used to assess the effect of road congestion on activity-travel scheduling behaviour. Road congestion not only induces a larger travel cost, but also reduces the time available for activity participation. Hence, the full cost of road congestion is widely greater than increased financial travel cost. The traditional trip-based models cannot be used to determine how the durations of activities, such as work and shopping, are affected by travel time. The financial cost benefit analysis based on trip-based models is thus biased.

The household MDP model has been used to examine the effect of intra-household interactions on the activity-travel schedule choice. Joint participation of certain non-compulsory activities, such as social visits and outdoor sports, provides a higher overall household utility because of the positive intra-household interaction. Thus, household members tend to participate in the same activity and at the same location. If the intra-household interaction is ignored, the travel demand at the locations of these activities will be underestimated. The intra-household interaction for other non-compulsory activities, such as cooking and cleaning, can be negative. Household utility is maximized by allocating this type of non-compulsory activity to one household member. As a result, the travel demand estimated by traditional trip-based models without consideration of negative intra-household interactions will be overestimated.

The maximum likelihood method is employed to calibrate the intra-household interaction coefficient and marginal utility function in the household's MDP model. The calibration method requires observations of household members' activity-travel decisions over time episodes. The activity-travel data required for calibration were generated from numerical experiments. The calibration method was tested and evaluated with these hypothetical data.

The achievements of this thesis on the development of a modelling framework are summarized and classified into the following five aspects.

- (1) In previous activity-based network equilibrium models, travellers are assumed to care only about the utility that can be obtained immediately and choose the activity that has the highest immediate utility. In the proposed MDP model, two types of within-day dynamics in activity-travel scheduling have been captured (Chapter 4). Firstly, to obtain the maximum overall utility in a day, the traveller not only cares about the utility derived from the current activity immediately but also is concerned about the utility that could be obtained during the remainder of the day. Secondly, the activity-travel decisions depend on contextual situations, such as time of day and location.
- (2) Each activity-travel schedule is individually treated and enumerated in the previous activity-travel scheduling models. The multi-dimension choices of activity types, destinations, timing and duration constitute many activity-travel schedules. These schedules impose a computational burden on the

implementation of the models. The proposed MDP model avoids activity-travel schedule enumeration by introducing a new structure (i.e., state) to represent activity-travel schedules (Chapter 4 and Chapter 5).

- (3) Intra-household interactions affect the individual household member's activitytravel scheduling. In the proposed household MDP model, the interactions between household member's activities are incorporated into a household utility function (Chapter 5) and examined for different activity types. As indicated above, the calibration of travel demand for locations of activities with nonzero interactions is biased in traditional models.
- (4) The allocation of activities within a household is also considered in the proposed household MDP model. Certain activities are compulsory for a specific household member, such as work and school, but the non-compulsory activities can be allocated to either household member (Chapter 5). Imposing these constraints on activity choice demonstrates the flexibility of the MDP framework.
- (5) Numerical methods are formulated and implemented to calibrate the MDP models. Notably, numerical experiments show that the log-likelihood function is *concave* over the domain of intra-household interaction coefficient  $\rho$  (Chapter 6). This property enables efficient and accurate calibration of coefficient  $\rho$ .

### 7.2 Limitations of the study and future research

Although this thesis covers many aspects of individual and household activitytravel scheduling, some questions deserve further investigations. The direction for future research is outlined as follows.

- (1) A potential extension of the individual and household MDP models is to consider day-to-day dynamics in activity-travel scheduling. The effect of certain activities can persist for multiple days and thus the activities participated in one particular day can influence the later activity-travel schedules (Arentze and Timmermans, 2009; Habib and Miller, 2008). For example, the goods purchased during a shopping trip can be adequate for a few days' usage. The probability of making another shopping trip on the next day will be very low. Another point to note is that activity-travel schedules on weekdays and weekends differ significantly. Compulsory activities, such as work and school are regular occurrences on weekdays, while some noncompulsory activities, such as physical exercise, are usually performed at the weekend.
- (2) In this thesis, all activities are categorized into two types: compulsory and noncompulsory. Based on the values of intra-household interactions, the noncompulsory activities are further categorized into activities with positive interactions and activities with negative interactions. It would be of interest to further categorize activities into even smaller groups based on their socioeconomic characteristics. In line with the contentions of Bradley and Vovsha (2004), the variation in intra-household interactions across activity types can then be examined at a finer level of detail.
- (3) This thesis only captures the intra-household interactions of a two-person household. In reality, there are different types of households, such as two fulltime workers with children, non-worker or part-time worker, and two retired persons (Vovsha et al., 2004). The difference between household members affects the household's activity-travel schedules. For example, young children cannot undertake grocery shopping by themselves. In addition, the number of feasible household states increases exponentially with the number of household

members. The computational burden is the major difficulty for modelling households with three or more individuals. However, approximate dynamic programming with interpolation can be employed to alleviate the computational burden as indicated by Keane and Wolpin (1994).

#### 7.2.1 Challenges in data collection

The limitations presented above are concerned with the modelling methodology. Another limitation is how to apply the proposed models in practice, especially, collecting the data required for model calibration. Travellers' activity-travel choices and geographic locations in a period are required for calibrating parameters of the MDP models. These fine-grained data cannot be readily collected by traditional travel survey methods. For example, travellers may not accurately recall the start time and duration of every activity in a household interview.

To extract more comprehensive information from traditional travel survey data, resampling techniques are adopted to generate data for model calibration (Miller and Roorda, 2003; Roorda et al., 2008). The activity generation is based on random draws of activities from observed probability distribution functions of activity frequency. Activity start time is then randomly drawn from a joint frequency-start time probability distribution function, conditional upon activity frequency. Finally, activity duration is randomly drawn from a joint start time-duration probability distribution function.

In addition to utilization of traditional travel survey data, applying the MDP models in real world also relies on advances in data collection technologies. For example, data mining technology can be used to extract activity locations and travel sequences from travellers' GPS trajectories (Zheng et al., 2010, 2009). As depicted in the left part of Figure 7.1, a GPS log is a collection of GPS points

 $P = \{p_1, p_2, ..., p_n\}$ . Each GPS point  $p_i$  contains latitude  $lat_i$ , longitude  $long_i$ and timestamp  $T_i$ .

1		Latitude	Longitude	Time	
	p <sub>1</sub>	Lat <sub>1</sub>	Long <sub>1</sub>	т <sub>1</sub>	Activity location
	p <sub>2</sub>	Lat <sub>2</sub>	Long <sub>2</sub>	т2	
					$P_2 \rightarrow P_6 \qquad P_7$
	p <sub>n</sub>	Latn	Long <sub>n</sub>	т <sub>n</sub>	P3

Figure 7.1 GPS points, trajectories and an activity location

The method for identifying activity location replies on the observation that an activity location is a geographic area where a traveller stays for some time to pursue activity. Thus, an activity location can be represented by a cluster of GPS points,  $L = \{p_a, p_{a+1}, p_{a+2}, ..., p_b\}$  satisfying the following conditions:

Distance 
$$(p_i, p_j) \le d_{thres}$$
  $\forall a \le i, j \le b$   
 $T_b - T_a \ge t_{thres}$ 

where  $d_{thres}$  and  $t_{thres}$  are two parameters defining the space and time thresholds.  $d_{thres}$  and  $t_{thres}$  can be assumed to be fixed or adaptively change with characteristics of GPS points. Further research is needed to determine the right values for  $d_{thres}$  and  $t_{thres}$  such that all the activity locations are accurately identified.

The start time and end time of the activity is simply  $T_a$  and  $T_b$ . The cluster of points is classified as a location of certain activity, such as shopping and work. The classification of cluster of points can be reported by the traveller in travel survey interview. However, if no travel survey interview is conducted, how to infer the corresponding activities for clusters of points is a major challenge in data processing.

# **Appendix I: MDP with temporal abstraction**

In the standard MDP model, the time episode is used to index the decision process. If the current choice is taken at t, the previous choice was made at t-1 and the next one at t+1. This formulation is convenient when the travel time between activity destinations is much shorter than the period represented by a time episode. In this case, travellers can make an activity-travel choice at every time episode.

If the travel time is longer than a time episode, travelling exclusively takes these time episodes and no activity can be scheduled during this period. For example, a traveller departs from home at 8:30am and arrives at the office at 9:00am. If each time episode represents a 10-minute period, the travel time between home and workplace equals to three time episodes. During this period, the traveller cannot make or implement any decisions. Thus, the time interval between consecutive choices should not be fixed at a constant value.

MDP models are conventionally based on a discrete time episode: the choice made at time t affects the state and utility at time t+1. There is no notion of a course of choice that persists over a period of time. Thus, a higher level of temporal abstraction is necessary to handle the activity-travel choice that lasts for a number of time episodes.

The choices that affect the states and utilities for multiple time episodes are termed *decisions*. They are also termed *options* in artificial intelligence literature (Sutton et al., 1999). A decision is viewed as a generalization of the primitive activity-travel choice made at each time episode. A decision *d* consists of three components: a policy for the primitive choices,  $\pi: S \times C \rightarrow [0,1]$ , a termination condition,  $\gamma: S \rightarrow [0,1]$ , and an initiation set  $\mathcal{I} \subseteq S$ .

For any state s if a decision is taken, activity-travel choices are selected according to  $\pi$  until the decision terminates according to  $\gamma$ . A decision satisfies the *Markov* property if its rule, termination condition and the initiation set depend only on the current state. In particular, a Markov decision executes as follows. First, the next choice  $c_k$  is selected according to the choice probability function  $\pi(s_k, \cdot)$ . The state then transitions from  $s_k$  to  $s_{k+1}$ . The decision either terminates with probability  $\gamma(s_{k+1})$ , or continues. If the decision continues, a new choice  $c_{k+1}$  is selected according to  $\pi(s_{k+1}, \cdot)$ . This process continues until the termination condition is reached. When the decision terminates, the traveller is able to make another decision. For example, a traveller makes a decision to go for shopping by taking the subway. The primitive choices may involve choosing a subway line, getting off the train, walking to the shopping mall and starting to shop. The decision terminates when the items on the shopping list are purchased or the traveller has been shopping long enough. The initiation set restricts engagement of shopping to states in which the previous activity has been completed and the shopping mall is open.

Given a set of decisions D, let D(s) denote the set of decisions in D that are available in state s according to the initiation set. D(s) resembles C(s) in the standard MDP model, in which C(s) denotes the set of primitive choices. Similarly, policies over decisions are defined. When initiated in state  $s_k$ , the Markov policy over decisions  $\mu: S \times D \rightarrow [0,1]$  selects a decision  $d_k \in D(s_k)$ according to the probability distribution  $\mu(s_k, \cdot)$ . The decision  $d_k$  determines primitive choices until it terminates in  $s_{k+\kappa}$ , where  $k+\kappa$  is a random time episode. A new decision  $d_{k+\kappa}$  is then selected in state  $s_{k+\kappa}$  according to  $\mu(s_{k+\kappa}, \cdot)$  and so on.

For any decision d and state s, let  $\mathcal{E}(d, s, k)$  denote the event of d being initiated in state s at time k. The total discounted utility of choosing d in state s is defined as:

$$r(s,d) = \mathbb{E}\Big[r_{k+1} + \beta r_{k+2} + \dots + \beta^{\kappa-1} r_{k+\kappa} \,|\, \mathcal{E}\big(d,s,k\big)\Big]$$

where  $k + \kappa$  is the random time at which *d* terminates. The state transition probability for choosing *d* in state *s* is then:

$$p(s'|s,d) = \sum_{\kappa=1}^{\infty} p(s',\kappa|s,d) \cdot \beta^{\kappa}$$

where  $p(s', \kappa | s, d)$  is the probability that the decision is initiated in s and terminates in s' after  $\kappa$  steps.

Given the utility and state transition probability of decision d, the Bellman equation for any Markov policy  $\mu$  in state s can be expressed as:

$$V^{\mu}(s) = \sum_{d \in D(s)} \mu(s, d) \cdot \left[ r(s, d) + \sum_{s'} p(s' | s, d) V^{\mu}(s') \right]$$

which is analogous to (3.1). The corresponding Bellman equation for the value of a decision *d* in state  $s \in \mathcal{I}$  is:

$$Q^{\mu}(s,d) = r(s,d) + \sum_{s'} p(s'|s,d) \sum_{d' \in D(s')} \mu(s',d') \cdot Q^{\mu}(s',d')$$

Finally, the optimal Bellman equations are as follows:

$$V^{*}(s) = \max_{d \in D(s)} \left\{ r(s,d) + \sum_{s'} p(s' | s,d) V^{*}(s') \right\}$$

$$Q^{*}(s,d) = r(s,d) + \sum_{s'} p(s'|s,d) \cdot \max_{d' \in D(s')} \{Q^{*}(s',d')\}$$

The policy over the primitive choices is defined as follows. If  $a_s \in A$  and  $e_s > 0$ , that is the current activity is not completed, the traveller chooses to continue the current activity in the next period  $t_s + 1$ . If  $a_s = \text{travel}$  and  $e_s > 0$ , that is the traveller has not arrived at the destination, the traveller must continue the trip in the next period  $t_s + 1$ . The initiation set includes the states that the current on-going activity or travelling is completed  $\mathcal{I} = \{s \mid e_s = 0\}$ .

If a decision is made, the associated policy over primitive choices is followed until the decision terminates. The corresponding termination rule is defined as:  $\gamma(s)=0$  for any  $s \notin \mathcal{I}$  and  $\gamma(s)=1$  for any  $s \in \mathcal{I}$ . This assumption is restrictive. Suppose that a traveller is heading for an activity destination. The traveller may cancel the planned activity and go to another location. The model formulation can be adapted to capture this type of interruption. Instead of following the decision until termination, the traveller can re-evaluate the decision at each step. That is comparing the utility of continuing with d, which is  $Q^{\mu}(s,d)$ , with the utility of making a new decision according to the policy  $\mu$ , which is  $V^{\mu}(s) = \sum_{d'} \mu(s,d') Q^{\mu}(s,d')$ . The MDP models in Chapter 4 and 5 do not

capture this type of interruption. However, the MDP models can be extended in this direction without difficulty.

# Appendix II: AMPL code for solving MDP models

### and calibrating model parameters

```
# Title: A Dynamic Markov Activity-Travel Scheduler
# Author: Xiong Yiliang <wlxiong@gmail.com> 2013
# Go to the NEOS Server (google "NEOS Server for Optimization").
# Click on "NEOS Solvers" and then go to "Nonlinearly Constrained
Optimization"
# You can use any of the constrained optimization solvers that take AMPL
input.
# AMPL Model File:
                     MarkovActv.mod
# AMPL Data File:
                      MarkovActv.dat
# AMPL Command File: MDPNonlinearEqn.run, JointMDPNonlinearEqn.run
                      MDPStateAction.run, JointMDPStateAction.run MLEMathProgEC.run, JointMLEMathProgEC.run
# control of debug logging
param debug_log;
# SET UP THE MODEL and DATA #
# Define and process the data
param T; # the equivalent minutes of a time episode
param H;
                # number of time episode in the data
               # the longest duration for a decision
param DH;
set TIME := 0..(H-1); # TIME is the vector of time episodes
param N := 2; # number of individuals in the household
set PERS := 1..N; # PERS is the index set of individuals
                        # number of activities, including HOME
# ACTV is the index set of activities
param M;
set ACTV := 1..M;
param HOME;
                         # define HOME activity
set WORK {n in PERS};
                        # define the work activity for each household member
# generated time serise data
param n1;
                          # a specific household member
param I;
                      # sample size
                         # sample IDs
set SAMPLE := 1...I;
param xt {SAMPLE, TIME};
                           # state: current activity
param dx {SAMPLE, TIME}; # decision: activity type
param dh {SAMPLE, TIME}; # decision: activity duration
param xt1 {SAMPLE, TIME}; # state: current activity of person 1
param xt2 {SAMPLE, TIME}; # state: current activity of person 2
param dx1 {SAMPLE, TIME}; # decision: activity type of person 1
param dx2 {SAMPLE, TIME}; # decision: activity type of person 2
# shortcuts for set union and set product
set AUW {n in PERS} := ACTV union WORK[n];
set AW1xAW2 := AUW[1] cross AUW[2];
set ALLACTV := ACTV union WORK[1] union WORK[2];
# Travel time varies over time of the day
param travelTime {TIME cross AW1xAW2};
param opening {ALLACTV}; # activity opening time
param closing {ALLACTV}; # activity closing time
# Declare the feasible states
param isFeasibleState {n in PERS, t in 0...H, j in AUW[n]} default 0;
param isFeasibleCoState {t in 0..H, (j1,j2) in AW1xAW2} default 0;
# Declare the feasible choices
param isFeasibleChoice {n in PERS, t in 0..H, j in AUW[n], k in AUW[n], h in
1..DH} default 0;
# Define the state space used in the dynamic programming part
# X is the index set of states
set X {n in PERS}:= {t in TIME, j in AUW[n]: isFeasibleState[n,t,j] == 1};
# XX is the set of composite states
set XX := {t in TIME, (j1,j2) in AW1xAW2:
    isFeasibleState[1,t,j1] == 1 and
```

```
isFeasibleState[2,t,j2] == 1 and
    isFeasibleCoState[t,j1,j2] == 1};
# DTRAVEL is the index set of travel decisions
set DT {n in PERS, (t,j) in X[n]} := {k in AUW[n], h in 1..DH:
    k != j and h == travelTime[t,j,k] and isFeasibleChoice[n,t,j,k,h] == 1};
# DA is the index set of activity decisions
set DA {n in PERS, (t,j) in X[n]} := {k in AUW[n], h in 1..DH:
    k == j and isFeasibleChoice[n,t,j,k,h] == 1};
# D is the union of sets of travel and activity decisions
set D {n in PERS, (t,j) in X[n]} := DT[n,t,j] union DA[n,t,j];
# DD is the set of composite decisions. To simplify the state transition,
# the activity durations of the component decisions should be the same.
set DD {(t,j1,j2) in XX} := {a1 in AUW[1], a2 in AUW[2], h in 1..DH:
     (a1,h) in D[1,t,j1] and (a2,h) in D[2,t,j2]};
# Parameters and definition of transition process
# Define discount factor. We fix beta since it can't be identified.
                        # discount factor
param beta;
# END OF MODEL and DATA SETUP #
# DEFINING STRUCTURAL PARAMETERS and ENDOGENOUS VARIABLES TO BE SOLVED #
# value of time
param VoT >= 0;
# theta: parameter of the logit choice model
param theta >= 0;
# intra-household interaction coeffcient for each activity
var rho {ALLACTV} >= -1.0, <= 1.0;
# true value
param rho0 {ALLACTV} >= -1.0, <= 1.0;
# calibrated value
param rho_ {ALLACTV} >= -1.0, <= 1.0;</pre>
# PARAMETERS OF CAUCHY DISTRIBUTION
# Is Cauchy distribution used ?
var IS CAUCHY;
# Activity parameters
var Um {PERS cross ALLACTV} >= 0, <= 5000;</pre>
var b {PERS cross ALLACTV} >= 0, <= 1440;
var c {PERS cross ALLACTV} >= 0, <= 600;</pre>
# True parameters
param UmO {PERS cross ALLACTV} >= 0, <= 5000;
param b0 {PERS cross ALLACTV} >= 0, <= 1440;
param c0 {PERS cross ALLACTV} >= 0, <= 600;</pre>
# Calibrated parameters
param Um_ {PERS cross ALLACTV} >= 0, <= 5000;
param b_ {PERS cross ALLACTV} >= 0, <= 1440;
param c_ {PERS cross ALLACTV} >= 0, <= 600;</pre>
# PARAMETERS OF BELL-SHAPED FUNCTION
# Activity parameters
param U0 {PERS cross ALLACTV};
param U1 {PERS cross ALLACTV};
param xi {PERS cross ALLACTV} >= 0, <= 1440;
param gamma {PERS cross ALLACTV};
param lambda {PERS cross ALLACTV};
# Marginal activity utility
param PI := 3.141592653;
var actvUtil {n in PERS, j in AUW[n], t in 0...H} =
    if IS CAUCHY == 1 then
         # Scaled Cauchy distribution
         if j == HOME and t >= H/2 then
              Um[n,j]/PI*(atan((t*T+T-(b[n,j]+1440))/c[n,j]) - atan((
t*T-(b[n,j]+1440) )/c[n,j]) )
         else
             Um[n,j]/PI*( atan( ( t*T+T-b[n,j])/c[n,j] ) - atan( ( t*T-
b[n,j])/c[n,j]) )
    else
         # Bell-shaped marginal utility function
         if j == HOME and t >= H/2 then
```

```
T * ( U0[n,j] +
                    gamma[n,j]*lambda[n,j]*U1[n,j] /
   ( exp( gamma[n,j] * (1440.0 - t*T - xi[n,j]) ) *
                          (1 + exp(-gamma[n,j]*(1440.0 - t*T - xi[n,j]))
)**(lambda[n,j]+1) ) )
         else
             T * ( U0[n,j] +
                    gamma[n,j]*lambda[n,j]*U1[n,j] /
    ( exp( gamma[n,j] * (t*T - xi[n,j]) ) *
                         (1 + exp(-gamma[n,j]*(t*T - xi[n,j]))
)**(lambda[n,j]+1) ));
# DECLARE EQUILIBRIUM CONSTRAINT VARIABLES
# The NLP approach requires us to solve equilibrium constraint variables
# Define initial values for EV
param initEV;
# Declare expected value of each component state
var EV {n in PERS, t in 0...H, j in AUW[n]} default initEV;
# Declare expected value of each composite state
var EW {t in 0..H, (j1,j2) in AW1xAW2} default initEV;
# Declare lower bound of EW
var lower {t in 0..H, (j1,j2) in AW1xAW2};
# Declare upper bound of EW
var upper {t in 0...H, (j1,j2) in AW1xAW2};
# END OF DEFINING STRUCTURAL PARAMETERS AND ENDOGENOUS VARIABLES #
# DECLARE AUXILIARY VARIABLES #
# Define auxiliary variables to economize on expressions
# Define the total discounted utility of pursuing activity j in time (t,
++h-1)
var sumActvUtil {n in PERS, (t,j) in X[n], (k,h) in DA[n,t,j]} =
    sum {s in 1..h} beta**(s-1) * actvUtil[n,k,t+s];
# Define the total discounted utility of traveling from j to k departing at
+
param sumTravelCost {n in PERS, (t,j) in X[n], (k,h) in DT[n,t,j]} =
    sum {s in 1..h} beta**(s-1) * T*VoT/60;
# Define the joint utility
var jointActvUtil {(t,j1,j2) in XX, (a1, a2, h) in DD[t,j1,j2]} =
    if al == a2 then
        sum {s in 1..h} beta**(s-1) *
sqrt(actvUtil[1,a1,t+s]*actvUtil[2,a2,t+s])
    else
         0.0:
# Define the utility of selecting decision (k,h)
var choiceUtil {n in PERS, (t,j) in X[n], (k,h) in D[n,t,j]} =
    if k == i then
          sumActvUtil[n,t,j,k,h]
    else
        - sumTravelCost[n,t,j,k,h];
# Declare the choice probability
var choiceProb {n in PERS, (t,j) in X[n], (k,h) in D[n,t,j]} =
    exp( theta * (choiceUtil[n,t,j,k,h] -
          beta**h * EV[n,(t+h),k]) -
theta * EV[n,t,j] );
# Define the joint decision utility
var jointChoiceUtil {(t,j1,j2) in XX, (a1, a2, h) in DD[t,j1,j2]} =
    if a1 == j1 and a2 == j2 and a1 == a2 then
        sumActvUtil[1,t,j1,a1,h] + sumActvUtil[2,t,j2,a2,h] +
    rho[a1] * jointActvUtil[t,j1,j2,a1,a2,h]
else if a1 == j1 and a2 == j2 then
        sumActvUtil[1,t,j1,a1,h] + sumActvUtil[2,t,j2,a2,h]
    else if al == j1 then
        sumActvUtil[1,t,j1,a1,h] - sumTravelCost[2,t,j2,a2,h]
    else if a2 == j2 then
        - sumTravelCost[1,t,j1,a1,h] + sumActvUtil[2,t,j2,a2,h]
    else
         - sumTravelCost[1,t,j1,a1,h] - sumTravelCost[2,t,j2,a2,h];
```

```
# Declare the joint choice probability
var jointChoiceProb {(t,j1,j2) in XX, (a1, a2, h) in DD[t,j1,j2]} =
    exp( theta * (jointChoiceUtil[t,j1,j2,a1,a2,h] +
                  beta**h * EW[t,a1,a2]) -
         theta * EW[t,j1,j2] );
# END OF DECLARING AUXILIARY VARIABLES #
# DEFINE OBJECTIVE FUNCTION AND CONSTRAINTS #
# TODO calculate profile of likelihood and draw 3D diagrams
# Define the objective: Likelihood function
    The likelihood function contains two parts
    First is the likelihood that the engine is replaced given time t state
in the data.
    Second is the likelihood that the observed transition between t-1 and t
would have occurred.
maximize likelihood0: 0;
maximize likelihood:
    sum {i in SAMPLE, t in TIME}
        if (t, xt1[i,t]) in X[n1] and (dx1[i,t], dh[i,t]) in D[n1, t,
xt1[i,t]] then
            log( choiceProb[ n1, t, xt1[i,t], dx1[i,t], dh[i,t] ] )
        else
            0.0:
maximize likelihood joint:
    sum {i in SAMPLE, t in TIME}
        if (t, xt1[i,t], xt2[i,t]) in XX and (dx1[i,t], dx2[i,t], dh[i,t])
in DD[ t, xt1[i,t], xt2[i,t] ] then
            log( jointChoiceProb[ t, xt1[i,t], xt2[i,t], dx1[i,t], dx2[i,t],
dh[i,t] ] )
        else
            0.0;
# Define the Bellman equation of the component MDP model
subject to Bellman Eqn {n in PERS, (t,j) in X[n]}:
    EV[n,t,j] = if card(D[n,t,j]) > 1 then
                     log( sum {(k,h) in D[n,t,j]}
                             exp( theta * (choiceUtil[n,t,j,k,h] +
                                            beta**h * EV[n,(t+h),k]) ) /
theta
                 else sum {(k,h) in D[n,t,j]}
                             (choiceUtil[n,t,j,k,h] + beta**h *
EV[n,(t+h),k]);
subject to Bellman_EqnH {n in PERS}:
    EV[n, H, HOME] = EV[n, 0, HOME];
# Define the Bellman equation of the composite MDP model
subject to Bellman Joint {(t,j1,j2) in XX}:
    EW[t,j1,j2] = if card(DD[t,j1,j2]) > 1 then
                     log( sum {(a1,a2,h) in DD[t,j1,j2]}
                             exp( theta * (jointChoiceUtil[t,j1,j2,a1,a2,h]
+
                                            beta**h * EW[t,a1,a2]) ) /
theta
                          sum {(a1,a2,h) in DD[t,j1,j2]}
                   else
                             (jointChoiceUtil[t,j1,j2,a1,a2,h] + beta**h *
EW[t,a1,a2]);
subject to Bellman JointH:
    EW[H, HOME, HOME] = EW[0, HOME, HOME];
# Define the Bellman equation for updating the lower and upper bounds
subject to Bellman Lower {(t,j1,j2) in XX}:
    lower[t,j1,j2] = log( sum {(a1,a2,h) in DD[t,j1,j2]}
                                 exp( theta *
(jointChoiceUtil[t,j1,j2,a1,a2,h] +
                                               beta**h * lower[t,a1,a2]) ) )
/ theta;
subject to Bellman LowerH:
    lower[H, HOME, HOME] = lower[0, HOME, HOME];
```

```
subject to Bellman_Upper {(t,j1,j2) in XX}:
    upper[t,j1,j2] = log( sum {(a1,a2,h) in DD[t,j1,j2]}
                                    exp( theta *
(jointChoiceUtil[t,j1,j2,a1,a2,h] +
                                                   beta**h * upper[t,a1,a2]) ) )
/ theta;
subject to Bellman_UpperH:
    upper[H, HOME, HOME] = upper[0, HOME, HOME];
# Define the lower and upper bounds for EW
subject to LowerBound {(t,j1,j2) in XX}:
    EW[t,j1,j2] >= lower[t,j1,j2];
subject to UpperBound {(t,j1,j2) in XX}:
    EW[t,j1,j2] <= upper[t,j1,j2];</pre>
# Symmetric parameters
subject to Symmetric_Um {j in ACTV}: Um[1,j] = Um[2,j];
subject to Symmetric_b {j in ACTV}: b[1,j] = b[2,j];
subject to Symmetric_c {j in ACTV}: c[1,j] = c[2,j];
```

```
# END OF DEFINING OBJECTIVE FUNCTION AND CONSTRAINTS
```

# References

- Abdelghany, A., Mahmassani, H.S., Chiu, Y.-C., 2001. Spatial Microassignment of Travel Demand with Activity Trip Chains. Transp. Res. Rec. 1777, 36–46.
- Abdelghany, A.F.A., Mahmassani, H.S.H.S., 2003. Temporal-Spatial Microassignment and Sequencing of Travel Demand with Activity-Trip Chains. Transp. Res. Rec. 1831, 89–97.
- Aguirregabiria, V., Mira, P., 2010. Dynamic discrete choice structural models: A survey. J. Econom. 156, 38–67.
- Arentze, T.A., Hofman, F., van Mourik, H., Timmermans, H.J.P., 2000. ALBATROSS: Multiagent, Rule-Based Model of Activity Pattern Decisions. Transp. Res. Rec. 1706, 136–144.
- Arentze, T.A., Timmermans, H.J.P., 2009. A need-based model of multi-day, multi-person activity generation. Transp. Res. Part B Methodol. 43, 251–265.
- Axhausen, K.W., Garling, T., 1992. Activity-based approaches to travel analysis: conceptual frameworks, models, and research problems. Transp. Rev. 12, 323–341.
- Baillon, J.B., Cominetti, R., 2008. Markovian traffic equilibrium. Math. Program. 111, 33–56.
- Ben-Akiva, M., Bierlaire, M., 1999. Discrete choice methods and their applications to short term travel decisions. Handb. Transp. Sci. 23, 5–35.
- Bhat, C.R., 1996. A hazard-based duration model of shopping activity with nonparametric baseline specification and nonparametric control for unobserved heterogeneity. Transp. Res. Part B Methodol. 30, 189–207.

- Bowman, J.L., Ben-Akiva, M.E., 2001. Activity-based disaggregate travel demand model system with activity schedules. Transp. Res. Part A Policy Pract. 35, 1–28.
- Bradley, M., Vovsha, P., 2004. A model for joint choice of daily activity pattern types of household members. In: Transportation. Washington, DC, pp. 545– 571.
- Charypar, D., Nagel, K., 2005. Q-Learning for Flexible Learning of Daily Activity Plans. Transp. Res. Rec. 1935, 163–169.
- Cirillo, C., Axhausen, K.W., 2010. Dynamic model of activity-type choice and scheduling. Transportation (Amst). 37, 15–38.
- Cirillo, C., Xu, R., 2011. Dynamic Discrete Choice Models for Transportation. Transp. Rev. 31, 473–494.
- Dafermos, S.C., 1980. Traffic Equilibrium and Variational Inequalities. Transp. Sci. 14, 42–54.
- Dial, R.B., 1971. A probabilistic multipath traffic assignment model which obviates path enumeration. Transp. Res. 5, 83–111.
- Eckstein, Z., Wolpin, K.I., 1989. The Specification and Estimation of Dynamic Stochastic Discrete Choice Models: A Survey. J. Hum. Resour. 24, 562–598.
- Ettema, D., Ashiru, O., Polak, J., 2004. Modeling Timing and Duration of Activities and Trips in Response to Road-Pricing Policies. Transp. Res. Rec. 1894, 1–10.
- Ettema, D., Bastin, F., Polak, J., Ashiru, O., 2007. Modelling the joint choice of activity timing and duration. Transp. Res. Part A Policy Pract. 41, 827–841.
- Ettema, D., Timmermans, H.J.P., 2003. Modeling Departure Time Choice in the Context of Activity Scheduling Behavior. Transp. Res. Rec. 1831, 39–46.

- Florian, M., Hearn, D.W., 1995. Chapter 6 Network equilibrium models and algorithms. In: M.O. Ball, T.L.M.C.L.M., Nemhauser, G.L. (Eds.), Handbooks in Operations Research and Management Science. Elsevier, pp. 485–550.
- Fourer, R., Gay, D.M., Hill, M., Kernighan, B.W., Laboratories, T.B., 1990. AMPL: A Mathematical Programming Language. Manage. Sci. 36, 519–554.
- Garling, T., Kwan, M.-P., Golledge, R.G., Garling, T., 1994. Computationalprocess modelling of household activity scheduling. Transp. Res. Part B Methodol. 28, 355–364.
- Golob, T., 2000. A simultaneous model of household activity participation and trip chain generation. Transp. Res. Part B Methodol. 34, 355–376.
- Habib, K., 2010. A random utility maximization (RUM) based dynamic activity scheduling model: Application in weekend activity scheduling. Transportation (Amst). 38, 123–151.
- Habib, K., Miller, E., 2008. Modelling daily activity program generation considering within-day and day-to-day dynamics in activity-travel behaviour. Transportation (Amst). 35, 467–484.
- Hatzopoulou, M., Miller, E.J., Santos, B., 2007. Integrating vehicle emission modeling with activity-based travel demand modeling - Case study of the greater Toronto, Canada, Area. Transp. Res. Rec. 29–39.
- Hensher, D.A., Greene, W.H., 2003. The Mixed Logit model: The state of practice. Transportation (Amst). 30, 133–176.
- Huang, H.-J., Lam, W.H.K., 2005. A Stochastic Model for Combined Activity/Destination/Route Choice Problems. Ann. Oper. Res. 135, 111–125.

- Jones, P. (Ed.), 1990. Developments in Dynamic and Activity Based Approaches to Travel Analysis (Oxford Studies in Transport). Avebury.
- Jonsson, R.D., Karlström, A., 2005. SCAPES a dynamic microeconomic model of activity scheduling. In: European Transport Conference.
- Kasturirangan, K., Pendyala, R.M., Koppelman, F.S., 2002. History Dependency in Daily Activity Participation and Time Allocation for Commuters. Transp. Res. Rec. 1807, 129–136.
- Keane, M.P., Wolpin, K.I., 1994. The Solution and Estimation of Discrete Choice Dynamic Programming Models by Simulation and Interpolation: Monte Carlo Evidence. Rev. Econ. Stat. 76, 648–672.
- Kitamura, R., 1984. Incorporating trip chaining into analysis of destination choice. Transp. Res. Part B Methodol. 18, 67–81.
- Kitamura, R., 1988. An evaluation of activity-based travel analysis. Transportation (Amst). 15, 9–34.
- Kitamura, R., Pas, E.I., Lula, C. V, Lawton, T.K., Benson, P.E., 1996. The sequenced activity mobility simulator (SAMS): an integrated approach to modeling transportation, land use and air quality. Transportation (Amst). 23, 267–291.
- Lam, W.H.K., Huang, H.-J., 2003. Combined Activity/Travel Choice Models: Time-Dependent and Dynamic Versions. Networks Spat. Econ. 3, 323–347.
- Lam, W.H.K., Li, Z.-C., Huang, H., Wong, S.C., 2006. Modeling time-dependent travel choice problems in road networks with multiple user classes and multiple parking facilities. Transp. Res. Part B Methodol. 40, 368–395.
- Lam, W.H.K., Yin, Y., 2001. An activity-based time-dependent traffic assignment model. Transp. Res. Part B Methodol. 35, 549–574.

- Li, Z.-C., Lam, W.H.K., Wong, S.C., Huang, H.-J., Zhu, D.-L., 2008. Reliability evaluation for stochastic and time-dependent networks with multiple parking facilities. Networks Spat. Econ. 8, 355–381.
- Lin, D.-Y., Eluru, N., Waller, S.T., Bhat, C.R., 2008. Integration of Activity-Based Modeling and Dynamic Traffic Assignment. Transp. Res. Rec. 2076, 52–61.
- Mahmassani, H.S., 1988. Some comments on activity-based approaches to the analysis and prediction of travel behavior. Transportation (Amst). 15, 35–40.
- Maruyama, T., Harata, N., 2005. Incorporating Trip-Chaining Behavior into Network Equilibrium Analysis. Transp. Res. Rec. 1921, 11–18.
- Maruyama, T., Sumalee, A., 2007. Efficiency and equity comparison of cordonand area-based road pricing schemes using a trip-chain equilibrium model. Transp. Res. Part A Policy Pract. 41, 655–671.
- McElroy, M.B., Horney, M.J., 1981. Nash-bargained household decisions: Toward a generalization of the theory of demand. Int. Econ. Rev. (Philadelphia). 22, 333–349.
- McFadden, D.L., Train, K., 2000. Mixed MNL Models for Discrete Response. J. Appl. Econom. 15, 447–470.
- McNally, M.G., 2000. The activity-based approach. In: Hensher, D.A., Button, K.J. (Eds.), Handbook of Transport Modelling. Pergamon, pp. 53–69.
- Miller, E., Roorda, M., 2003. Prototype Model of Household Activity-Travel Scheduling. Transp. Res. Rec. 1831, 114–121.
- Oppenheim, N., 1995. Urban Travel Demand Modeling: From Individual Choices to General Equilibrium. Wiley-Interscience.

- Ouyang, L., Hom, H., Kong, H., Li, Z.-C., Huang, H.-J., 2010. A Network Equilibrium Model for Scheduling Daily Activity-Travel Patterns. Sci. Technol. 1–22.
- Ouyang, L., Lam, W., Li, Z., Huang, D., 2011. Network User Equilibrium Model for Scheduling Daily Activity Travel Patterns in Congested Networks. Transp. Res. Rec. 2254, 131–139.
- Puterman, M.L., 1994. Markov Decision Processes: Discrete Stochastic Dynamic Programming, 1st ed. John Wiley & Sons, Inc., New York, NY, USA.
- Ramadurai, G., Ukkusuri, S., 2011. B-Dynamic: An Efficient Algorithm for Dynamic User Equilibrium Assignment in Activity-Travel Networks1. Comput. Civ. Infrastruct. Eng. 26, 254–269.
- Ramadurai, G., Ukkusuri, S. V, 2010. Dynamic User Equilibrium Model for Combined Activity-Travel Choices Using Activity-Travel Supernetwork Representation. Networks Spat. Econ. 10, 273–292.
- Roorda, M.J., Miller, E.J., Habib, K., 2008. Validation of TASHA: A 24-h activity scheduling micro simulation model. Transp. Res. Part A Policy Pract. 42, 360–375.
- Rust, J., 1987. Optimal Replacement of GMC Bus Engines: An Empirical Model of Harold Zurcher. Econometrica 55, 999–1033.
- Rust, J., 1988. Maximum Likelihood Estimation of Discrete Control Processes. SIAM J. Control Optim. 26, 1006.
- Rust, J., 1994. Chapter 51 Structural estimation of markov decision processes. Handb. Econom. 4, 3081–3143.
- Sheffi, Y., 1985. Urban Transportation Networks. Prentice-Hall, Englewood Cliffs, NJ.

- Singh, S., Cohn, D., 1998. How to dynamically merge Markov decision processes. Adv. Neural Inf. Process. Syst. 1057–1063.
- Su, C.-L., Judd, K.L., 2008. Constrainted Optimization Approaches to Estimation of Structural Models.
- Supernak, J., 1992. Temporal utility profiles of activities and travel: Uncertainty and decision making. Transp. Res. Part B Methodol. 26, 60–76.
- Sutton, R.S., Precup, D., Singh, S., 1999. Between MDPs and semi-MDPs: A framework for temporal abstraction in reinforcement learning. Artif. Intell. 112, 181–211.
- Train, K., 2003. Discrete Choice Methods with Simulation. Cambridge University Press, Cambridge.
- Vovsha, P., Petersen, E.R., Donnelly, R., Trb, 2004. Impact of intrahousehold interactions on individual daily activity-travel patterns. In: 83rd Annual Meeting of the Transportation-Research-Board. Washington, DC, pp. 87–97.
- Wen, C.-H.H., Koppelman, F.S., 2000. A conceptual and methological framework for the generation of activity-travel patterns. Transportation (Amst). 27, 5–23.
- Winston, G.C., 1982. The Timing of Economic Activities: Firms, Households and Markets in Time-Specific Analysis, Cambridge Books. Cambridge University Press.
- Winston, G.C., 1987. Activity choice: A new approach to economic behavior. J. Econ. Behav. Organ. 8, 567–585.
- Zhang, J., Kuwano, M., Lee, B., Fujiwara, A., 2009. Modeling household discrete choice behavior incorporating heterogeneous group decision-making mechanisms. Transp. Res. Part B Methodol. 43, 230–250.

- Zhang, J., Timmermans, H.J.P., Borgers, A., 2005. A model of household task allocation and time use. Transp. Res. Part B Methodol. 39, 81–95.
- Zheng, Y., Xie, X., Ma, 2010. GeoLife: A Collaborative Social Networking Service among User, Location and Trajectory. Data Eng. 33, 32–40.
- Zheng, Y., Zhang, L., Xie, X., Ma, W.-Y., 2009. Mining interesting locations and travel sequences from GPS trajectories. Proc. 18th Int. Conf. World wide web - WWW '09 791.